LOOKING BACKWARD, MOVING FORWARD: THE EXPERIENCES OF INDO-FIJIAN IMMIGRANT WOMEN IN CALIFORNIA

A Thesis

Presented to

The Faculty of the Department of Anthropology

San José State University

In Partial Fulfillment

Of the Requirements for the Degree

Master of Arts

by

Ambrita Nand

December 2015

© 2015

Ambrita Nand

ALL RIGHTS RESERVED

The designated Thesis Committee Approves the Thesis Titled

LOOKING BACKWARD, MOVING FORWARD: THE EXPERIENCES OF INDO-FIJIAN IMMIGRANT WOMEN IN CALIFORNIA

by

Ambrita Nand

APPROVED FOR THE DEPARTMENT OF ANTHROPOLOGY

SAN JOSÉ STATE UNIVERSITY

December 2015

Dr. Chuck Darrah Department of Anthropology

Dr. James Freeman Department of Anthropology

Dr. Roberto Gonzalez Department of Anthropology

ABSTRACT

LOOKING BACKWARD, MOVING FORWARD: THE EXPERIENCES OF INDO-FIJIAN IMMIGRANT WOMEN IN CALIFORNIA

This study helps address gaps in knowledge concerning the lives of Indo-Fijian immigrant women in California and offers a space for their voices to be heard. The subsequent chapters investigate the lives of five Indo-Fijian immigrant women and their experiences upon migrating to Modesto, California. Using a qualitative research approach, data were collected through participant-observations, semi-structured in-depth interviews and informal conversations. The data are presented as anthropological silhouettes, a form of life-writing (the recording of events and experiences of a life), which explores each individual woman's experience with life in Fiji to her eventual migration and transition to life in California. The study reveals heterogeneity amongst the women's experiences and perspectives as well as commonalities that arise in their collective experiences as Indo-Fijian immigrant women residing in the city of Modesto. Overall, the anthropological silhouettes reveal that migration has led to shifts in the women's identities and their prescribed gender roles. Furthermore, despite some of the challenges that came with immigrating, the women have experienced social, political and economic mobility since arriving to California. All five women have accepted the United States as their adopted homeland, and as a result, have no plans of re-migrating to Fiji.

ACKNOWLEDGMENTS

This thesis could not have been written without the support and guidance of many people. First and foremost, I would like to thank all the extraordinary women who took time out of their busy lives to share their stories and memories with me. I would also like to express my appreciation for Dr. Chuck Darrah, Dr. James Freeman, and Dr. Roberto Gonzalez. Collectively, your insights and expertise were invaluable to my research. Finally, I would like to thank my family and friends for giving me constant strength and encouragement during my journey through graduate school. Thank you Mom, Dad, Monica, Nimita, and Ian for always being there for me.

CONTENTS

| Chapter 1: Introduction | 1 |
|--|----|
| Purpose and Significance of the Study | 3 |
| Organization of the Thesis | 4 |
| Chapter 2: Historical Context. | 5 |
| Introduction | 5 |
| Part 1: Fiji's Colonization. | 5 |
| European Contact and Colonization | 5 |
| Contracting Indian Laborers | 7 |
| Part 2: Indentured Period. | 9 |
| Plantation Life and Plight of Indian Laborers | 9 |
| Cultural Fragmentation and the Birth of the Indo-Fijian Identity | 12 |
| The Abolition of the Indentured System | 14 |
| Part 3: Post-Indentured Years | 16 |
| Seeking Political and Economic Representation in Fiji | 16 |
| Land Issues | 18 |
| Fiji's Independence | 19 |
| Women's Movement and Women's Rights | 20 |
| Part 4: Four Coups in Twenty Years | 21 |
| First Coup: May 1987 | 21 |
| Second Coup: October 1987 | 22 |
| Third Coup: May 2000 | 24 |

| | Fourth Coup: December 2006 | 26 |
|------|---|----|
| | Indo-Fijian Emigration From Fiji | 26 |
| | The Years After the Coup and Fiji's Current Situation | 27 |
| | Chapter Summary | 28 |
| Chap | oter 3: Methodology | 30 |
| | Introduction | 30 |
| | Understanding Women's Perspectives: Feminist Anthropology | 30 |
| | Qualitative Research Design | 33 |
| | Research Site | 34 |
| | Selection of Participants. | 36 |
| | Description of the Research Participants. | 37 |
| | Krishna | 37 |
| | Malti | 38 |
| | Priya | 39 |
| | Asha | 39 |
| | Sheela | 40 |
| | Methods for Data Collection. | 41 |
| | Participant Observations | 41 |
| | In-Depth Interviews | 42 |
| | Informal Conversations | 43 |
| | Reflexive Approach | 44 |
| | Approaches to Data Analysis and Presentation | 46 |

| Transcription | 46 |
|--|----|
| Thematic Analysis | 46 |
| Anthropological Silhouettes | 47 |
| Limitations to the Methodology | 48 |
| Chapter Summary | 49 |
| Chapter 4: Anthropological Silhouettes | 51 |
| Introduction | 51 |
| Anthropological Silhouette: Krishna | 51 |
| Growing Up in Nausori | 51 |
| Schooling Years: Primary School | 52 |
| Secondary School and Abuse | 54 |
| Misfortune of the Bride | 56 |
| Abandonment | 60 |
| Coming to America and the American Workforce | 62 |
| Second Arrange Marriage | 63 |
| Purchasing a Home and Raising Children in California | 64 |
| Anthropological Silhouette: Malti | 66 |
| Growing Up in Nadi | 66 |
| Schooling Years | 66 |
| Moving to Suva, Children, and Remembering the First Coup | 67 |
| From Suva City to "Fairyland" | 69 |
| Raising Children in California | 70 |

| Converting to Christianity | 71 |
|--|----|
| Proud to Be an American | 72 |
| Visiting Fiji | 73 |
| Anthropological Silhouette: Priya. | 74 |
| Early Life in Fiji | 74 |
| Marriage, Family, and Tensions From the Coup | 75 |
| Moving to Modesto | 76 |
| Employment and Buying a House | 76 |
| Return to Fiji | 77 |
| Citizenship: Becoming American | 78 |
| A Visit to the Motherland India | 79 |
| Raising Grandchildren | 79 |
| Anthropological Silhouette: Asha | 80 |
| Growing Up in Suva, Fiji | 80 |
| Leaving Home at 12 | 81 |
| Growing Up in South San Francisco | 83 |
| Marriage and American-Born Children | 84 |
| Returning to the Homeland | 85 |
| Anthropological Silhouette: Sheela | 85 |
| Early Life, Memories of Fiji | 85 |
| The Move to Colusa County | 86 |
| Schooling | 87 |

| The Move to Modesto and High School Years | 87 |
|--|-----|
| Marriage | 88 |
| Employment | 88 |
| Child-Rearing | 89 |
| Converting to Catholicism | 90 |
| Sponsoring Family During the Coups | 90 |
| No Return, Yet | 91 |
| Chapter Summary | 92 |
| Chapter 5: Discussion. | 93 |
| Introduction | 93 |
| Push Factors: Economic and Political. | 93 |
| The Nature of Migration: Family Sponsorship | 95 |
| Reuniting and Separating. | 96 |
| Social, Political and Economic Mobility | 97 |
| Maintaining the Indo-Fijian Identity and Not Just Immigrants but | |
| Transmigrants | 99 |
| Women's Resiliency and Immigration as Success | 101 |
| Implications and Contributions of the Research. | 102 |
| References Cited | 107 |
| Appendix A: Interview Consent Form. | 115 |
| Appendix B: Interview Protocol and Sample Questions | 117 |

Chapter 1: Introduction

Nestled in the South Pacific between the equator and the tropic of Capricorn, a vast archipelago of more than 330 islands forms the breathtaking nation of Fiji. With postcard-perfect scenery, pristine white sand beaches and friendly locals, this island paradise attracts an impressive number of visitors each year. Ironically, while tourists arrive to Fiji, many of those born on native Fijian soil have been coerced from their country of birth. However, those coerced from Fiji are not the typical faces found on the travel brochures and Internet sites for the nation. Instead, those who have departed represent the lesser-known segment of Fiji's segregated plural society, the Indo-Fijians. The Indo-Fijians are the direct descendants of the *girmitiyas*, or indentured laborers from India, who first arrived to Fiji in 1879 to work for the sugar industry.

Since the arrival of the *girmitiyas*, Indo-Fijians have struggled through unremitting social, political and economic marginalization in Fiji. Not only did the community bear through 37 years of indenture, Indo-Fijians were also victimized during Fiji's four military coups. The coups generated countless human rights violations against Indo-Fijians, leading to their eventual exodus to four main overseas destinations including Australia, Canada, New Zealand and the United States of America (B. Lal 2003). Since their departure from Fiji, little research has been conducted on Indo-Fijians in their new host societies. At present, only three serious studies on Indo-Fijians immigrants exist. They include Norman Buchignani's (1980) study of Indo-Fijians in

Canada, Carmen Voigt-Graf's (2001) study of Indo-Fijians in Australia, and Monica Devi's (2012) study on Indo-Fijians in California's East Bay.

Coupled with the lack of knowledge on Indo-Fijian immigrants, Indo-Fijian women constitute one of Fiji's most vulnerable groups, one that has endured the highest levels of marginalization not only because of their ethnicity but also their gender. The marginalization of Fiji's Indian women has been documented since the indentured period. The rare collection of literature on these women depicts them as docile, inferior, powerless, promiscuous, and victims of both physical and sexual violence.

Understanding these women's predicaments is truly difficult because as Indo-Fijian historian Brij Lal (1985) explains, their voices are "absent in the written records, where for the most part, their faces are shrouded by a veil of dishonor drawn *by* men" (1985:142). Shireen Lateef (1987), a noted Indo-Fijian anthropologist, also comments on the androcentric biases surrounding the lives of these women:

The vast majority of commentaries mostly written by men, are imbued with derogatory comments about [Fiji's Indian indentured] women. Few have bothered to discuss in any detail women's working conditions or attempted to analyze indentured women's lives from a perspective other than a male centered one. This bias pervades much of the historical data. (1987:4)

Regrettably, the lack of Indo-Fijian women's voices persists today, not to mention an overall lack of knowledge about their lives in Fiji or of those living abroad. Indo-Fijian women continue to remain largely invisible and silent in Fiji as well as in their new host societies.

Purpose and Significance of the Study

Due to the lack of research on Indo-Fijian immigrants in their native environment as well as in their new host societies, combined with the severe underrepresentation of Indo-Fijian women's voices and perspectives throughout Fiji's history, I attempt to contribute in filling these gaps in knowledge by exploring the lives of Indo-Fijian immigrant women in the United States. Thus, the study has two objectives; the first is to provide a space for Indo-Fijian women's voices to be heard, while the second objective is to learn about the gendered experience of Indo-Fijian emigration to the United States.

Due to the growing Indo-Fijian diaspora in Modesto, California, the city provides a good place to start exploring the lives of Indo-Fijian immigrant women in the United States.

With various Indo-Fijian shops, temples and cultural associations around Modesto, the city has allowed the community to flourish and preserve their unique cultural heritage.

Using a qualitative research approach, I investigated the experiences of five Indo-Fijian immigrant women by learning about their lives in Fiji and their eventual migration and transition to life in Modesto, California. Data for the study were collected through the means of participant observations, semi-structured in-depth interviews and informal conversations. The data are then presented in the form of anthropological silhouettes, a form of life-writing (the recording of events and experiences of a life), which allowed for exploring both the individual and collective experiences and perspectives of the women. The anthropological silhouettes provide a glimpse into the women's lives prior to migration and after migration, showcasing both differences and commonalities amongst their experiences and perspectives. Ultimately, this research makes important contributions to the study of Indo-Fijian emigration to the United States while providing a space for Indo-Fijian women's voices to be heard.

Organization of the Thesis

This chapter introduces the purpose and significance of the study. Chapter 2 provides a historical context to Fiji's Indo-Fijian community. The objective of this chapter is to acquaint readers to the geneses of the Indo-Fijian identity as well as the social, political and economic history of the Fiji Islands. By having this knowledge, one can begin to understand the factors that directly and indirectly shaped the lives of the women in the study and led to their eventual migration to California. Chapter 3 provides a discussion on the methodology utilized for the study. Specifically, this chapter details the framework that guided the study, the selection of participants, methods for data collection and analysis, and limitations of the methodology. Next, chapter 4 presents the anthropological silhouettes, which chronicle the lives of five Indo-Fijian immigrant women residing in Modesto, California who appear under the pseudonyms Krishna, Malti, Priya, Asha, and Sheela. Finally, chapter 5 provides a discussion of the themes that emerged from the anthropological silhouettes as well as the implications and contributions of the study.

Chapter 2: Historical Context

Introduction

To better understand the experiences of the women in the study, it is first important to consider the larger historical context of Fiji, their native birthplace. Thus, this chapter introduces the origins of the Indo-Fijian identity as well as the social, political and economic factors in Fiji that directly and indirectly shaped their lives and led to their eventual migration to California. This chapter also exposes the limited knowledge concerning the lives of Indo-Fijian women and their vulnerability throughout Fiji's history.

To enhance readability, this chapter is divided into four parts. First, part 1 details European contact and colonization, contracting Indians to Fiji, and the origins of the laborers. Next, part 2 focuses on the indentured period, 1879-1919, and reviews plantation life, the plight of the laborers, and the eventual abolition of the indentured contract. Part 3 examines the post-indenture years and Indo-Fijian efforts for social, political and economic representation in Fiji. Finally, part 4 focuses on Fiji's four military coups, Indo-Fijian emigration, the years after the coups, and Fiji's current situation.

Part 1: Fiji's Colonization

European Contact and Colonization

The picturesque scenery of Fiji and its lush environment have attracted settlers from near and afar for centuries. The original inhabitants of Fiji are said to have arrived approximately 3,500 years ago from the islands of Melanesia (Devi 2012:13). Besides

the original inhabitants whose descendants are now considered the indigenous people of Fiji, Europeans also journeyed to the islands. The earliest mention of European contact in Fiji was in 1643 when Dutch explorer Abel Adam sighted the archipelago. Since then, European contact was amplified upon the discovery of sandalwood (1800-1814), a fragrant wood used to barter with the Chinese, and with the sea cucumber trade, which lasted from the 1820s to the 1850s (B. Lal 1992:9). However, it was the discovery that Fiji provided the ideal environment for growing sugar cane that led Europeans to stay on the islands. British traveler H. Stonehewer Cooper (1879) describes in his manuscript *Fiji and its Prospects* the potential for the crop in Fiji:

Among tropical products the cultivation of the sugar cane, for the manufacture of sugar, Fiji holds a prominent place. Every visitor to Fiji (even from countries in which the sugar cane is the only plant cultivated in any extent) cannot fail being struck with the size, the healthy appearance and the rapidity of growth of the sugar cane in Fiji . . . By the foregoing it will be noted that the climate of Fiji is by nature well adapted for growing the sugar cane, and so is the soil . . . The extent and richness of these lands in conjunction with a climate extremely favorable for growing and maturing the sugar cane, ought to make all well-wishers of Fiji long for the time when sugar will be made in and *exported by the hundred thousands tons, and to the value of millions of pounds sterling*. (1879:14-15)

With the vision of high profits, the British government quickly grasped control of the islands, and on October 10, 1874, Chief Cakobau agreed to the "Deed of Cession," which placed Fiji under British colonial rule (Mayer 1963:8). Under the leadership of Sir Arthur Gordon, Fiji's first governor, various policies were designed including that land remain under Fijian ownership and that indigenous Fijians be exempt from participating in the sugar plantation economy itself. Also, the British Crown developed a separate Fijian administration, which allowed them to govern the indigenous Fijians indirectly

(Premdas 1995:10-12). In addition to these policies, strict segregation of laborers from the indigenous Fijian population was imposed. It was by no accident that the two groups were separated: It was merely the colonial strategy of "divide-and-rule" that allowed the British to maintain control in Fiji.

Contracting Indian Laborers to Fiji

With the exemption of indigenous Fijians from participating in the sugar industry, a viable workforce was needed in Fiji. Former governor of Mauritius and Trinidad, Sir Arthur Gordon, who had witnessed the success of contracting laborers from India, suggested contracting Indian laborers for Fiji's sugar plantations. His recommendation led to the arrangements of an indentured contract or *girmit* (agreement in English) for Fiji, which harnessed laborers to five years of work related to the cultivation and manufacturing of agricultural goods. Laborers were required to work nine hours on the weekday, and five hours on Saturday, with Sundays and holidays off (Lal 2004a:6). And if laborers continued on for an additional five years, they were offered a free passage back to India. However, the indentured system would become highly unpopular with a re-indenturing rate of 5 percent (Gillion 1962).

On May 15, 1879, 463 *girmitiyas* arrived aboard the *Leonidas* to the island of Viti Levu (Nicole 2011:101). With 87 voyages from 1879 to 1916, a total of 60,965 indentured Indians eventually arrived at Fiji. Approximately 13,696 of these laborers were female, and overall there were unequal ratios of women to men; for every 40 women, there were 100 men (Lateef 1987:1). Nearly 80 percent of the laborers were from North India, particularly from the United Provinces (B. Lal 2004b:56). However,

most recruits were from the settled plains district of Bihar, while a small number of laborers came from the Central Provinces, Punjab, and Madras (B. Lal 2004b:51-52). Also, the laborers represented a "fair cross-section of Indian rural society" and over 260 social groups consisting of Hindu castes/sub-castes, Muslims, and tribal groups entered Fiji, with nearly 80 percent identifying as Hindus (B. Lal 2004b:99). While men made up the majority of the recruits, women, children and entire families also made the journey across the *kala panis* or the dark waters as the *girmitiyas* dubbed it (B. Lal 1998).

The majority of the laborers worked for the Colonial Sugar Refining (CSR) Company of Sydney, with their first mill established along the Rewa River in Nausori. Later, more mills were established across the island of Viti Levu in Ba, Lautoka and Penang (Gillion 1956:139). While plantation owners made ample profits as predicted by Stonehewer (1879), the laborers were under-compensated and over-worked by their employers. Indentured men received one shilling per day while women earned only nine pennies a day (B. Lal 2012:182).

Various factors contributed to the emigration of Indians including the lack of employment, extreme poverty, droughts and famines in India (B. Lal 1983:55, 62). Although these were some of the push factors that led many Indians to migrate to Fiji, others arrived under false pretenses; Indians were often enticed with propaganda. An Indian laborer named Totaram Sanadhya described in his rare account *My 21 Years in Fiji* how numerous Indians, like himself, were deceived and taken to Fiji:

The *arkati* (recruiter) explained things to people there [in the half-way house], 'Look brothers, the place you will work you will never have to suffer any sorrows. There will never be any problems there. You will eat

a lot of bananas and a stomach-full of sugar cane, and play flutes [symbolic of Lord Krishna] in relaxation.' (Totaram 1991:34)

Alluding to the false image of a carefree life as exquisite as that of Lord Krishna was difficult for some to resist. When fabricated stories did not work, recruiters resorted to more coercive measures. For example, historian Adrian Mayer (1963) describes in *Indians in Fiji* the difficulties of recruiting laborers such as women:

It was difficult to persuade married women to emigrate, however; and when the recruiters would not fill their quotas with such willing emigrants as widows and destitutes, for whom a life in Fiji promised a fresh start, they resorted to trickery and abduction. The recruitment of women, in fact, was even less of a 'free contract' than that of men, and the proportion of women was barely maintained. (1963:14)

Nonetheless, those who never intended on setting foot in Fiji would find themselves in the most precarious situation.

Part 2: Indentured Period

Plantation Life and the Plight of Indian Laborers

Plantation life was treacherous for the Indian laborers, and Fiji quickly became referred to as *narak*, the Hindi word for "hell" (B. Lal 2004a:87). The *girmitiyas* lived in deplorable, cramped living conditions called plantation lines, which usually consisted of eight small rooms on each side. The rooms were a minuscule ten feet by seven (after 1908 it was increased to ten by twelve feet), where three single men were assigned to one of the rooms or one man, a woman, and no more than two children to a room (Ali 1977:1783). With no doors or windows, the cramped quarters of the plantation also lacked proper ventilation (Ali 1977:1783). The closed and confined spaces of the plantation lines led to the proliferation of illnesses and diseases amongst the laborers.

Sores, ulcers, yaw, dysentery and tuberculosis quickly spread from one laborer to another (Ali 1977:1783). Although medical care was included in the indentured contract, the quality of care was poor. Laborers were often taken to unsanitary hospitals led by uncertified European doctors (Gillion 1962:106). Laborers were also severely malnourished; their diet became known as the "scurvy diet" for its lack of nutrition. For the first six months of their stay, food was rationed, and after six months, laborers were required to find their own food. The meager wages earned by the laborers were used to pay for the rations (a fourpence was deducted from their wage for each ration), which were distributed weekly. Eventually, rations were increased, which was necessary for the back-breaking work of the sugar industry (Gillion 1962:105-106). But to make matters worse, plantation owners often disregarded the nine-hour workday and overseers and plantation owners' systematically overtasked laborers, making it difficult for laborers to earn their pay without also having to work longer hours than stated in the Immigration Ordinance (Gillion 1962:110).

But the plight of the laborers was felt most through the systematic abuse and violence they endured. Indian laborers were often struck and beaten by plantation overseers. As for women laborers, they not only faced physical violence but sexual abuse. The disproportionate ratio of women to men made the indentured women easy targets for abuse by both their colonial masters and by indentured men. Between 1890 and 1919, it was reported that 68 indentured women were murdered in Fiji (Hyam 1990:94). According to Brij Lal (2001), sexual jealousy was perceived as the culprit behind the murders:

Indian indentured women thus stood accused in the eyes of their own community as well as those of the official world, carrying the double backpack of racism and sexism. The widely held, though empirically unsupported, perception of them as morally lax, profligate individuals made the women an easy target for malicious gossip and innuendoes. More seriously, it gave the *sirdars* (Indian foremen) and overseers the license to treat women with little respect and to view them simply as objects of sexual gratification. Not surprisingly, the indentured women also became convenient scapegoats for all the ill of the indentured system. (2001:198-199)

The double burden of being Indian and a woman took its toll on the lives of indentured women.

Eventually, the plight of the laborers led to high rates of suicide amongst the indentured Indians. In fact, Fiji had the highest suicide rates in all of the indentured colonies (Lal 1993:187). Totaram Sanadhya revealed some of the motives that led the laborers to take their own lives:

Overseers commit outrages against us whenever they like. Many of our brothers there [Fiji] make a noose and hang themselves, from fear of hard work, and from fears of jail, and the blows of the overseers. Not many days ago several Madrasis at a plantation in Navua hanged themselves for this reason. (Nicole 2011:163)

Robert Nicole further elaborates in his book *Disturbing History: Resistance in Early Colonial Fiji* that "the combination of fear, pain, displacement, hopelessness and helplessness created a profoundly alienating situation that in some circumstances made death more attractive than life" (Nicole 2011:164). Disturbingly, Fiji's high suicide rate continues today at endemic levels with the majority of the victims being young Indo-Fijian women (Radio Australia 2012).

Cultural Fragmentation and the Birth of the Indo-Fijian Identity

The culture and traditions that the laborers brought with them from India faced major transformation in Fiji. According to Brij Lal (2004a), "in that new environment [Fiji], old habits, patterns of thought and association, and understanding about their world and their place were shaken" (2004a:12). The patterns and process of change can be best illustrated by examining the caste system, marriage and gender relations amongst the laborers. Perhaps, the greatest and most evident change for the girmitivas was the disintegration of the caste system, which had lost its practical relevance in the everyday life of the migrants (B. Lal 2004a:12). Regardless of their caste position, laborers were made to live, eat and sleep within the same quarters. Each laborer was *individually* contracted to the plantation, and their pay was based on the amount of work he or she accomplished, rather than according to social status (B. Lal 2004a:12). Even if the laborers wanted to sustain the caste system in Fiji, they could not as they lacked the cultural resources to do so. The disintegration of the caste system was advantageous to the lower caste Indians who became equals to their higher caste counterparts. According to Subramani (1979), "While indentured enslaved the Indian laborer, it also, paradoxically, released him from an old, static order . . . it was the first stage in the transformation of the feudal Indian into an individual" (1979:x).

The institution of marriage also underwent drastic changes during indenture. For both Hindus and Muslims alike, marriages traditionally "took place within a narrow, restricted circle prescribed by customs." This, however, could not be preserved due to the lack of women in Fiji (B. Lal 2004a:13). Instead, the disproportionate sex ratio led to

inter-caste and cross-caste marriages. The institution of marriage was also further complicated since colonial law failed to recognize Hindu and Muslim marriages as illegitimate. This discrepancy led to the exploitation of women. For example, some indentured men agreed to their wives providing sexual services to a small number of men for a fee. Also, some fathers were known to sell their daughters (B. Lal 2004a:13). Overall, the plantation system undermined a *stable* family life, but marriages continued and families were still raised (B. Lal 2004a:14). Brij Lal (2004a) states, "The symbolic importance and comforting effect of the family during a time of turmoil in the life of the community cannot be overstated" (2004a:14). Although not well understood, families played an important role in the lives of the *girmitiyas* and their descendants.

In terms of gender relations amongst the indentured laborers, women occupied a marginal position. Their male counterparts blamed them for murders, suicides, and even the high infant mortality rate. Some indentured men viewed indentured women as a "temporary convenience to be discarded on returning to India" (B. Lal 2004a:14). Furthermore, men assessed women's roles on the plantation in stereotypes. For instance, Brij Lal (2004a) explains:

Some men measured them [women] against the ideal of Sita, the paragon of Hindu womanhood, who gave up everything to accompany her husband, lord Rama into exile. The ideal Indian woman accepted her fate without complaint, glorified the virtues of motherhood, deferred to male authority, and above all, worshipped her husband. On the plantations, men sought to reassert the patriarchal structure of agrarian Indian society. (2004a:14)

Thus, patriarchy was reinstated and functioned amongst the Indian community in Fiji, which still functions today.

Moreover, the languages, cultural and religious practices that the laborers brought with them from India were also altered. For example, the various Hindi dialects of the laborers, in combination with Fijian and English, resulted in a unique dialect called Fiji Hindi or "Fiji Baat" (Mangubhai and Mugler 2006:32-33); it is the primary language still spoken by the majority of Indo-Fijians in Fiji. Furthermore, since the majority of the laborers had little to no experience in the practice of Brahmanical Hinduism in India, many traditional ceremonies were not observed in Fiji. Others, however, such as *Diwali* and Holi continued but in simplified versions, while plays based on sacred texts such as the Ramayana were also reenacted. In fact, the Ramayana became the most sacred text amongst the Indo-Fijians Hindus because of its core themes of exile, suffering, struggle and eventual return (Voigt-Graf 2008:88). Moreover, Indian foods, spices, music and clothing were also revived in Fiji and flourished with the arrival of Gujarati and Punjabis entrepreneurs (Voigt-Graf 2008:89). Nevertheless, the result of the fragmentation led to the eventual birth of a distinct hybrid identity (Bhabha 1994, Gilroy1993, Gounder 2011:37-38), the Indo-Fijian identity.

The Abolition of the Indentured System

The molestation and abuse of indentured women provided a catalyst for the abolition of the indentured system in Fiji. One event in particular, concerning a 21-year-old indentured woman named Kunti, received special attention in both Fiji and India. In 1908, Kunti migrated from Lakhanpur village in Gorakhpur, India to Fiji with her husband to work on the sugar cane plantations. On April 10, 1913, Kunti was sent to work in an isolated banana field where an overseer approached her. The overseer

attempted to rape Kunti, but she resisted him and ran towards a river. Out of desperation, Kunti jumped into the river but was quickly rescued by an Indo-Fijian boy. Kunti's story received significant attention in India after being published in the national Indian newspaper the *Bharat Mitra* (B. Lal 1985:55-57). Also gaining wide attention was a violent incident against an indentured woman named Naraini, who was asked by an overseer to return to work only three days after giving birth to a dead child. After refusing to come to work, Naraini was beaten by the overseer to the point of unconsciousness. The incident made its way to Fiji's Supreme Court, but the overseer was found innocent (B. Lal 2004b:129).

The plight of indentured women could not be hidden. According to European missionary Miss Hannah Dudley, their quandaries were engraved on their faces:

Life on the plantations alters their demeanor and even their very faces. Some look crushed and broken-hearted, other sullen, others hard and evil. I shall never forget the first time I saw indenture women when they were returning from a day of work. The look on those women's faces haunts me. (Nicole 2011:200)

It was more than apparent that life as an indentured servant in Fiji was a life full of sorrow, pain and suffering. In 1914, Indian activist Mahatma Gandhi sent British missionary C.F. Andrew to investigate the indentured system in Fiji. After discovering the inhumane treatment of Indian laborers, Andrew and his colleague W.W. Pearson devised a report and called for the immediate abolition of the indentured system in Fiji (Mayer 1963:21). They wrote:

We cannot forget our first sight of the coolie 'lines' in Fiji. The looks on the faces of those men and the women alike told one unmistakable tale of vice. The sight of young children in such surroundings was unbearable to us. And again and again . . . we saw the same unmistakable look. It told

us of a moral disease which was eating into their heart and life of the people. (Nicole 2011:173)

In Fiji, and in other sugar colonies, it was evident that there was little difference between indentured servitude and slavery (Tinker 1974). So, on January 2, 1920, the remaining contracts were withdrawn and the indentured system abolished (Mayer 21:1963). Still, a cheap labor supply was needed in Fiji. As a result, the colonial government encouraged Indo-Fijians to remain in Fiji as free settlers. Of the 60,696 indentured laborers that arrived to Fiji, only 24,000 eventually returned to India (B. Lal 1998:2). B. Lal (1998) further elaborates on Indian settlement in Fiji:

The majority stayed on [in Fiji], trapped by the promise of a better life, dread of a long journey, the fear of rejection by family and friends of those who had broken caste taboos, and by the encouragement of a government keen to develop a local pool of labor supply. Time passed and memories of India faded as people formed new, cross-caste relationships and developed new attachments to their adopted homeland. (1998: 2)

With no immediate efforts by the colonial government to repatriate the laborers, Indo-Fijians adopted Fiji as their new homeland and began the long process of healing and uplifting a community that endured through the 37-year period of indenture in Fiji.

Part 3: Post-Indentured Years

Seeking Social, Political and Economic Representation in Fiji

With the cancellation of the indentured contract and a substantial community to accommodate, Indo-Fijians began seeking social, political and economic representation in Fiji. While the majority of Indo-Fijians continued as cane growers, either as independent farmers or tenants of CSR (B. Lal 1992:75), others began finding employment outside the sugar industry. As for Indo-Fijian women, their life drastically

changed. Instead of working for the sugar industry, they were confined to the home. This was difficult for the women who had lost their independent incomes. Indo-Fijian women had no choice but to marry Indo-Fijian men who had acquired land through leases from indigenous Fijians (Lateef 1987:5). Despite the cancellation of the contract, economic inequality and social discrimination continued against the Indo-Fijian community. Indo-Fijian activists began advocating for change through forms of nonviolent resistance. For example, on January 15, 1920, a group of Indo-Fijians employed at Suva's Public Works Department walked off their job demanding an increase in their wages and the same benefits offered to their Fijian and European counterparts. After hearing news about the strike, Indo-Fijian employees in Rewa also left their jobs. The strike lasted two months until an innocent Indo-Fijian activist was killed by a police officer (B. Lal 1992:81).

Although Indo-Fijians were met with resistance, they continued to advocate for the betterment of their community. For instance, in 1921 a group of Indo-Fijians held a sugar strike in Western Viti Levu. This time, Indo-Fijian tenants of the CSR estates demanded "better wages, specified work hours, adequate housing, medical and pension benefits, educational facilities for their children, and a small plot of land in which they could keep milk cows" (B. Lal 1992:82). To little surprise, CSR refused to cooperate and officers were called to intimidate the Indo-Fijian farmers (B. Lal 1992:82).

As resistance through strikes and other forms of non-violent protest continued, the community also worked towards other goals such as non-communal voting (Trkna 2008:34). The Indo-Fijians continued to face opposition for their efforts towards social equality and justice in Fiji. In 1923, a group of Europeans residing in Fiji declared they

"will resist, and will also encourage the native Fijians to resist with all means at their disposal, the contemplated attempt to admit Indian residents of Fiji to be the body of politic or to granting them any measure- however small- of political status" (B. Lal 1992:87). Pitting indigenous Fijians and Indo-Fijians against one another, European colonists sought to exclude Indo-Fijians from participating in the government. However, the resilient Indo-Fijian community began organizing themselves prior to the post-indentured era. For example, religious, social and political organizations such as the Arya Samaj, Sangama TISI, and Sanatam Dharm united Indo-Fijians in their efforts to uplift the community. They made significant contributions to the social, political and cultural growth of the community (Voigt-Graf 2010:1112-1113). For instance, the Arya Samaj, Sangama TISI, and Sanatam Dharm opened schools for Indo-Fijians. (B. Lal 1992:159). They understood that education was a vehicle for escaping poverty (Nicole 2011:178, 181).

Land Issues

To truly secure their future progress in Fiji, Indo-Fijians focused a great deal of attention on issues regarding land tenure. With over 83 percent of land belonging to indigenous Fijians and the rest to the British Crown, Indo-Fijians remained landless in Fiji. After the cancellation of the indentured contract, unused native land was leased to farmers since it could not be sold privately, but the leases were often short-term and non-renewable. But, in the late 1960s and 1970s, the Indo-Fijian community was temporarily subdued when land leases were increased to thirty years (B. Lal 2003). These leases began expiring in 1997 and issues concerning land tenure resurfaced at the turn of the

century as indigenous Fijians refused renewal of land leases to their Indo-Fijian tenants, leaving them no option but to remain impoverished in Fiji's deplorable squatter settlements (Bitter Harvest 2001).

Fiji's Independence

In 1965, the United Nations began conversations with the British Crown concerning Fiji's independence. These conversations eventually led to Fiji's independence on October 10, 1970. Under the new Constitution and Bill of Rights, Indo-Fijians were safeguarded and protected as equals to their indigenous Fijian counterparts. But the tide turned quickly, when members of the Fijian Nationalist Party (FNP) became uncertain about their political and economic future in Fiji. The FNP expressed their dissatisfaction for the new constitution, which treated Indo-Fijians as their equals. The disgruntled leader of the FNP, Sakeasi Butadroka, publically stated in 1975:

The time has arrived when Indians or people of Indian origin in this country be repatriated back to India and that their travelling expenses back home and compensation for their properties in the country be met by the British Government. (V. Lal 1990:19)

Desiring a Fiji without Indo-Fijians, Butadroka advised the descendants of the hardworking *girmitiyas* to relocate to India. For Butadroka, the growth Indo-Fijian community served as a threat. By 1976, the Indo-Fijian population of 295,000, accounted for 50 percent of Fiji's population (Kelly 1991:2), and many had entered in business and improved their economic positions in Fiji. Added to this, many Indo-Fijians were also attending secondary schools, and according to Helen Tavola "the secondary school students of the 1960s would be the professionals of 1970s and the leaders of the

1980s" (Tavola 1992:27). For instance, leaders such as Jai Ram Reddy emerged from this era and fought vehemently for Indo-Fijian rights in Fiji (B. Lal 2010).

Women's Movement and Women's Rights

The years from 1976 to 1985 were declared by the United Nations as the "Decade for Women: Equality, and Development." This decade inspired the creation of Fiji's first official feminist organization The Fiji's Women's Rights Movement (FWRM) (FWRM 2006:2). On April 10, 1986, fifty-six women of diverse backgrounds gathered at Des Voeux road in Suva to rally against "laws and policies which discriminated against women or that failed to adequately address their concerns" (FWRM 2006:1). This meeting led to the creation of the FWRM, which focused on improving the socioeconomic and political statuses for *all* women of Fiji. The FWRM tackled major issues such as rape, domestic violence and conditions of women workers in the garment industry (FWRM 2006:3). Some of the most significant leaders of the FWRM included women of Indo-Fijian descent such as Shamina Ali, a founding member of the FWRM and the coordinator of the Fiji's Women Crisis Center (FWCC), and FWRM lawyer Patricia Imrana Jalal.

By 1987, the FWRM voiced their concerns in Fiji's politics. Progress was on the horizon for the women's movement, especially with the newly elected Fiji Labor Party (FLP), which had promised a department of women's affairs, an anti-sex discrimination bill, equal pay legislation, minimum wage regulations for garment and domestic workers and funding for the FWCC. However, the coups ensued the same year, leading to the postponement of such promises. The effects of the coups led the FWRM to become

polarized based on ethnic lines. Many indigenous Fijians left the movement but some would later return to support the organization (FWRM 2006:15). In the years to follow, the FWRM concentrated much of their efforts on Fiji's return to democracy. They believed that if Fiji was to make future progress, then an unbiased constitution was needed to protect the fundamental rights for *all* people.

Part 4: Four Coups in Twenty Years

First Coup: May 1987

Many others besides Butadroka foreshadowed Fiji's impending progress as a nation. In 1963, Adrian Mayer wrote in *Indians in Fiji*:

Since the desire of some people to deport Indo-Fijians is impractical, two alternatives remain. One is that the present communal separation will continue, with chronic tension and later, perhaps, periodic outbreaks of violence as economic and political pressure . . . This would hamper political evolution, would drive away outside investors, and would bring misfortune and suffering for many people. (1963:135)

Having witnessed the separation between indigenous Fijians and Indo-Fijians himself, Mayer's predictions became a reality, and the small nation soon earned its disconcerting title as "Coup-Coup Land" (Robbie 2001, Revington 2011:98). Following the April 1987 parliamentary election, Fiji witnessed its first military coup led by Lieutenant Colonel Sitiveni Rabuka. Panicked by the fact that the multi-ethnic labor-led coalition the Fiji Labor Party (FLP) had come into power, Rabuka retaliated. On May 14, 1987, Rabuka and his men hijacked and detained the fairly elected Prime Minister Timcoi Bavadra, forcing him out of office (Lal 2004a:323). According to Rabuka, the coup was necessary for combatting "the Indian design for political domination" (Sharpham 2000:10, Trkna 2008:37). The propaganda of Indian domination led many indigenous Fijians to support

the overthrow, believing that the 1970 Constitution did not adequately protect their interest.

As soon as Rabuka took power, the army restricted coverage of newspapers, journalists were muzzled, and a military sensor controlled broadcasts in Fiji (Dobell 2008:126). Vijendra Kumar, an Indo-Fijian editor of Fiji's national newspaper the Fiji Times, described suffering from "four years of harassment, intimidation, and outright threats" (B. Lal 2004a:323). He explained, "I could no longer honestly and without fear discharge my professional duties. I applied to migrate to Australia" (B. Lal 2004a:323). Several Indo-Fijian academics were also imprisoned while others were abducted and brutalized (Lal 2004a:327). Actively silenced during these periods of unrest, Indo-Fijians were deprived of their fundamental human rights such as their freedoms to speech, expression and association. To make matters worse, Indo-Fijians could not rely on the police for help as "bribery and corruption had become a way of life" during the coup (B. Lal 2004a:327). The general feeling amongst the Indo-Fijian community was that of fear, insecurity and uncertainty for their futures in Fiji. The results of the chaos led many to contact their relatives abroad in hopes of fleeing Fiji. Thus began the exodus of Indo-Fijians from their nation of birth to four main destinations abroad including Australia, Canada, New Zealand and the United States of America.

Second Coup: October 1987

Less than half a year after Fiji's first military coup, Rabuka waged Fiji's second coup. Although the Fijian Supreme Court considered the first coup unconstitutional, negotiations between the dismissed government and the Alliance Party prompted Rabuka

to revolt once more. On September 23, 1987, an agreement was reached between the factions and resulted in the Deuba Accord. The Dueba Accord sought to create national unity through representing both parties under the leadership of the Governor-General. However, Rabuka was dissatisfied by the accord and waged a second coup, proclaiming himself as leader of Fiji, and Fiji a Republic (Rabuka 2012:13).

By 1990, Rabuka and his cabinet created a new constitution, which favored indigenous Fijians. But, the new constitution was short-lived for its "racist provisions," and by 1997 a new constitution, which safeguarded Indo-Fijians was revealed to the public (Trkna 2008:38). By this time, however, many skilled and educated Indo-Fijians had already departed Fiji. Between 1987 and 1996, an estimated 5,100 Indo-Fijian professionals emigrated, of whom 21 percent were architects, engineers, and related technicians, 15 percent accountants, 31 percent teachers, 12 percent medical, dental, veterinary, and related workers, and 21 percent other professionals (B. Lal 2003). B. Lal states, "The impact of their loss on Fiji is visible and acutely felt, particularly in health and education. Once reasonably self-sufficient in medical personnel, Fiji now imports doctors from overseas. And there is a growing shortage of science and mathematic teachers" (B. Lal 2003). The loss of the skilled and educated Indo-Fijians greatly impacted the developing nation. B. Lal (2003) further comments on the high cost of emigration of Fiji, noting the nation loses on average FJD\$44.5 million annually due to loss of skill, re-training new appointees, and delayed appointments. As for unskilled and poverty-stricken Indo-Fijians, they had little choice but to endure through the coups until Fiji returned back to "normalcy" (Trnka 2008). Between 1987 and 1996, 50,050 emigrated from Fiji, the annual average being 5,005 (B. Lal 2003).

Third Coup: May 2000

In Mid-March of 2000, an article appeared in Fiji's national paper *The Fiji Times* entitled "Migration Is Key," where Rabuka offered his perspective on the future of Indo-Fijians and the coming expiry of land leases. Rabuka predicted the potential for increased racial divide on the basis of territories and jobs. Although he was certain that society would live through the ethnic division, he yearned for Indo-Fijians to permanently depart Fiji. Foreshadowing Fiji's third coup, Rabuka retorted:

My other hope is that Indians will migrate. We tighten controls, then Fiji is no longer attractive to Indian settlers as it has been over the last 120 years. Maybe they will slow down their immigration rate, probably increase their emigration rate, thereby having a natural decline in their numbers, to a level that would be manageable. When I'm talking about a manageable level I am talking about the tolerance threshold of the Fijians. Now it is beyond their tolerance level and that is why they are reacting, not because of the numbers but of what the Indians can do with the numbers. If they gain control of the professional life, the economic life, and the political life of this country with the numbers they have, then the tolerance level, the numbers level will have to be greatly reduced. And we won't do that by the Butadroka-kind policy, forced migration, but by circumstances- driven migration. (Sharpham 2000:316-317)

Rabuka's rant failed to recognize the fact that the majority of Indo-Fijians were the direct descendants of the *girmitiyas*, and that the Indo-Fijian community in Fiji was actually decreasing. Between 1987 and 1990, approximately 44,000 people emigrated from Fiji; the majority of the emigrants were Indo-Fijians (Asian Pacific Migration Research Network 2014).

Fiji's third coup followed shortly after the May 1999 parliamentary election when voters elected Mahendra Chaudhry, the leader of Fiji's Labor Party as the nation's first Indo-Fijian prime minister (Trkna 2008:8). As Fiji's newest prime minister, Chaudhry hoped to tackle issues surrounding land tenure, but his term was short-lived because in May of 2000, a group of renegade soldiers led by Fijian businessman George Speight entered Parliament and kidnapped Chaudhry and members of his cabinet, holding them hostage for 56 days (Robie 2001). In the meantime, an interim military government led by military Commodore Frank Bainimarama was set up, and Ratu Josefa Iloilo was declared president. But by November, the High Court declared the interim government illegal and reinstated the 1997 constitution (Robbie 2001). Although restored, the chaos and capture of Chaudhry prodded even more Indo-Fijians to emigrate from Fiji. The instability had once again instilled fear within the Indo-Fijian community. According to anthropologist Susana Trnka (2008), postcolonial Fiji had never seen the levels of violence that were experienced in the 2000 coup:

Much of the civilian violence was directed against Indo-Fijians in their homes, businesses, and properties. It was fueled by a racialized anti-Indian rhetoric that promotes images of Indo-Fijians as *vulagi*, or foreigners, who had usurped the rights of the *taukei*, or indigenous Fijians, to govern it. (2008:3)

Additionally, Trkna (2008) discussed the fear of rape of Indo-Fijian women:

Another attribute that appears in many of the rural violence narratives, frequently mentioned by both Indo-Fijian men and women, is of the Fijian assailant as a sexual predator. Rapes and threats of rape occurred during the coup, but even in cases where the threat was not explicitly made, there was a widespread assumption among many Indo-Fijians that when young Fijian men came into violent contact with Indo-Fijian women, there was the likelihood of rape . . . Many Indo-Fijians fears of potential victimization were primarily concentrated on women of childbearing age

or a few years younger, who were seen as the most likely targets of sexual violence. The result was that women from this age group were subject to specific concern and safeguarding. (2008:160)

Real or imagined, the victimization of Indo-Fijians caused apprehension and anxiety within the community. The third coup pushed out another 24,000 Indo-Fijians from Fiji, and for those who could not leave the islands resorted to other means. An alarming 200 Indo-Fijians committed suicide after the third coup, the highest of any single community in the world (Mishra 2007:42).

Fourth Coup: December 2006

In December of 2006, Fiji experienced its fourth coup. This coup was a direct retaliation to the 2000 coup and because of two bills under consideration at the time: the Qoliqoli Bill, which gave Fijian proprietary ownership of foreshore land and the Reconciliation, Tolerance, and Unity Bill, which granted individual amnesties to coup perpetrators and supporters (Ratuva 2007:38). Dissatisfied with the bills, Commodore Bainimarama overthrew the government on December 5, 2006, and declared Qarase's government corrupt for directing racially discriminatory policies against Indo-Fijians. He then transferred executive power back to Ratu Josefa Ilolio and declared himself interim Prime Minister of Fiji (Fraenkel 2007:420).

Indo-Fijian Emigration From Fiji

With four coups within a time span of 20 years, relentless issues over land and discriminatory treatment, Indo-Fijians were coerced from their country of birth. In 1987 the total population of Fiji was 721, 594 (World Bank 2015), and with the series of coups in Fiji, a total of 70,000-80,000 Indo-Fijians emigrated from Fiji after 1987 (B. Lal

2012:180). Altogether, even prior to the coups, over a third of the Fiji's Indo-Fijian population emigrated from the islands, or approximately 150,000 individuals (Naidu 2001). Geographer Manoranjan Mohanty, estimated that between 2000 and 2006 Fiji lost over 3,800 professional and technical specialists, which represented more than half of the nation's stock of middle high level workers (Miller 2008:387). The results of the exodus were devastating to Fiji. Brij Lal put it simplest, exclaiming:

Three coups in 13 years, two in 1987, and one in 2000, have dealt a severe blow to the island's economy, shaken investor confidence, strained race relations already frayed in an ethnically divided society, and corrupted the institutions and practices of good governance. Perhaps the most important consequence in the long term has been the emigration of the country's best and brightest to greener pastures in North America and Australasia, draining the small island nations of skill and talent it can ill-afford to lose. The tide of emigration is not likely to ebb anytime soon. (2003)

With the departure of Indo-Fijian professionals from Fiji, the nation now grapples with the demand for skilled laborers.

The Years After the Coups and Fiji's Current Situation

From 2006 to 2009, Fiji remained under the military dictatorship of Frank Bainimarama. In 2009, Bainimarama rescinded Fiji's third constitution and declared martial law in Fiji. Then in 2012, Bainimarama announced consultations for a new constitution in Fiji and promised for a fair election to be held in 2014. Interestingly, although Bainimarama had voiced early on that he would not run for the position of prime minister in the upcoming election, he ended up running as a proponent of the Fiji First Party and won by an overwhelming majority. Currently, Bainimarama is hoping to achieve democracy in Fiji through the new constitution.

Interestingly enough, perhaps the greatest irony behind the Indo-Fijian experience is that the same Indo-Fijians who were once forced from their nation of birth, are *now* being called upon to return to Fiji. On August 11, 2015, less than a year after being elected prime minister, Frank Bainimarama issued an apology to individuals who escaped from Fiji to Canada after the 1987 and 2000 coups. Echoing B. Lal, Bainimarama stated that their departure led to Fiji's brain drain and ultimately prevented the islands from decades of development. He also admitted that Fiji's citizens had suffered tremendously because of "certain selfish elements said they didn't belong." With multiple citizenships now allowed in Fiji, these former citizens are being urged to reconnect with their country of birth (Radio New Zealand International 2014).

Chapter Summary

This chapter provides historical background for a consideration of the experiences of Indo-Fijian women. Although the experiences of those women were, of course, idiosyncratic, they also reflect a set of relatively uniform conditions under which they made personal decisions about their futures; granted most of their rationales were not explicitly stated at the time and must be inferred. The chapter showcases Indo-Fijian women's marginality in Fiji and also the lack of information about their lives where they remain invisible and silenced throughout the nation's history. It also reveals something about the conditions affecting Indo-Fijian men as well, conditions that indirectly and directly affected the women's experiences and decisions. But despite common conditions, Indo-Fijian women responded in a variety of ways so that we must be cautious about inferring a "typical" person or outcome. Subsequent chapters will explore

these responses by examining a sample of women living in Modesto, California. But first, it is important to establish how the research was undertaken.

Chapter 3: Methodology

Introduction

This chapter presents the methodology used to investigate the lives of five Indo-Fijian immigrant women residing in Modesto, California. First a brief synopsis of feminist anthropology is offered as it provided a guiding framework for the study. Next, the chapter provides a discussion on the research design and explains the selection of participants, methods for data collection and data analysis. Finally, the chapter reviews the limitations of the methodology used for the study.

Understanding Women's Perspectives: Feminist Anthropology

While men are equally important to a holistic understanding of the Indo-Fijian immigrant experience, this study focuses exclusively on the lives of Indo-Fijian immigrant women because little is known about their experiences with migration to California. Also, since Indo-Fijian women have remained largely silent and invisible throughout history, I sought to engage in research that would bring women from the margins to the center. This led me delve into feminist anthropology, which ultimately stimulated my interest in the women's lives.

Nevertheless, feminist anthropology, which first emerged as a sub-discipline of anthropology in the mid 1970s (Lewin 2006:1), is an approach that critically evaluates relationships between men and women, focuses on understanding the nature of power, and connects political economy and culture. According to feminist anthropologist Louise Lamphere (2006), "feminist anthropology has always been about critique- critical appraisal of the social structures and cultural ideologies that shape women's lives and

reformulation of the theoretical apparatus that anthropologist have used to understand these structures and cultural notions" (2006:x).

Although feminist anthropology emerged as a formal sub-discipline of anthropology in the 1970s, feminist anthropology has been in the making since the late 19th century, and according to Kamala Viswesweran (1994), evolved over three major overlapping temporal waves. The first wave of feminist anthropology is rooted in the suffragist movements of the Progressive era. The goal of this wave, which extended from 1880 to 1920, was to begin recording women's roles in culture (Aggrawal 2000:17). However, men conducted most ethnographic research at this time, and the knowledge they produced on women was usually taken from the perspectives of male informants. As a result, feminist anthropology materialized as a reaction to the androcentric bias in the discipline (Lamphere 1996:488).

Spanning from 1920 to 1980, the second wave focused on the academic sphere. During this phase, the notions of sex and gender were separated from one another. It was realized that the definition of gender varied from culture to culture (Lamphere 1996:488). During this wave, researchers relied heavily on materialistic perspectives and Marxist theories. According to feminist anthropologist Gayle Rubin (1975), Marxist theory appealed to feminist anthropologists because no theory accounted "for the oppression of women in its endless variety and monotonous similarity, cross-culturally and throughout history with anything like the explanatory power of the Marxists theory of class oppressions" (1975:160). Furthermore, during second wave feminism, gender not only emerged as a unit of analysis, but it also became a "foundational lens of analysis"

(Aggrawal 2000:17). Texts such as Second Sex [1952] by Simone de Beauvoir and Feminine Mystique [1963] by Betty Friedan inspired anthropological works such as Women, Culture and Society [1974], edited by Michelle Rosaldo and Louise Lamphere, and Toward an Anthropology of Women [1975], edited by Rayna Rapp. Also during this phase, the concept of universal sisterhood emerged, which was the view that women all over the world shared similar experiences. It was assumed that anthropology "done by women on women would erase the prejudices of masculinist science and that the shift from analytical ethnographies to experience-based narratives would signify authentic discourse on women" (Aggrawal 2000:17). However, this view was discredited by anthropologists such as Rosaldo (1980), who were dissatisfied with anthropological projects that tended to cast other women in the "image of ourselves undressed" (1980:392). Ravina Aggrawal (2000) notes Judith Stacey (1988) who describes "betrayal" as inherent in ethnography and that discords between the expectations, methods, and writing of feminism, at best produced "partially feminist ethnography" (Aggrawal 2000:18). Added to this, Marilyn Strathern (1987) described the "awkward" relationship between anthropology and feminism. Aggrawal (2000) describes that Strathern (1987) believes that feminism helped to expose anthropology's patriarchal dominance, which subjected all women to be "otherized," while on the other hand, anthropology mocked feminism's "illusions of empathetically identifying with the subjection of other by pointing to feminists' common origin within a Western scientific paradigm" (Aggrawal 2000:18).

By the 1980s, third wave feminist anthropology made headway with its radicalism and innovation. The major focus of this phase was to deconstruct the idea of universal sisterhood. Feminist anthropologists began acknowledging other categories of difference that intersected with gender such as race, ethnicity, class and sexuality (Aggrawal 2000:18). Other critiques ensued during third wave feminist anthropology. Some feminists became skeptical of fieldwork, which they associated with "unilateral Western hegemony" and "otherizing." Aggrawal (2000) describes that according to Kamala Visweswaran (1994), fieldwork "propagates a doctrine of separating the notion of home as the ultimate domain of intellectual activity from the image of the field as a transformative but temporary space, generally confined to remote or peripheral locales. Instead, Visweswaran (1994) recommends that anthropologists engage in homework rather than fieldwork. She suggests taking more seriously the "knowledge, humanity, and history" when studying a culture, and for anthropologists to reflect on themselves, so they may better understand how their biases affect their research (Aggrawal 2000:19).

Qualitative Research Design

A qualitative research design was used to investigate the women's lives and was appropriate for the exploratory study. Although there are diverse traditions in conducting qualitative research, which vary from discipline to discipline, all qualitative research designs assume "non-objectivity and advocates contextualized knowledge and a multiperspective holistic approach" (Berger 2004:30). Berger (2004) states in her book *Immigrant Women Tell Their Stories* that "Understanding the complex world of lived experience from the point of view of those who live it and seeing through their lens is the

core of qualitative research" (2004:30). Since the study was a preliminary exploration on Indo-Fijian immigrant women in the United States, and also because I was interested in learning about the perspectives and experiences of the women from their own voices, a qualitative research was the most appropriate since data consisted of mainly words to which meaning would be ascribed. Thus, a quantitative research design, which seeks to quantify data into numbers or use statistical methods, was simply inappropriate for meeting the study's objectives.

The qualitative research methods used for the study included participantobservations, semi-structured interviews and informal conversations, which allowed for
gaining an in-depth understanding of the women's perspectives and experiences as
immigrants residing in Modesto, California. The process of qualitative research was also
flexible and allowed for the direction of the study to change as it proceeded. For
example, the original study focused on the lives of first-generation and second-generation
Indo-Fijian women residing in Modesto, California. However, since this was the first
foray into the field, I decided it was appropriate to first establish research on firstgeneration Indo-Fijian immigrant women, which will then allow for proposing future
research questions on the unique diaspora, and ultimately on the lives of secondgeneration Indo-Fijians.

Research Site

Historically, California has been a popular destination for immigrants around the world, and this still remains the case today. In the last half of the century, California has received more immigrants than any other state, and these immigrants are also ethnically

and racially diverse. In 2014, the U.S. Census revealed California to be more multicultural than ever before with Hispanics comprising 38.6 percent of the population, followed by 14.4 percent Asians, 6.5 African-Americans, 1.7 percent Native Americans and 0.5 percent Native Hawaiians and other Pacific Islanders (U.S. Census: California Quick Facts 2014).

Moreover, major cities in California such as Los Angeles and Sacramento are recognized for their ethnic diversity. TIME magazine acknowledged Sacramento as the most ethnically diverse community in California, where people are said to live side by side more successfully than in other cities (Stoghill and Bower 2002). Modesto, which is an hour drive south of Sacramento, is also a richly diverse city. Modesto is comprised of 35.5 percent Hispanics, 6.7 percent Asians, 4.2 percent African-Americans, and 1.0 percent Native Hawaiians or Pacific Islanders (U.S. Census: Modesto Quick Facts 2014). With a population of more than 200,000, Modesto attracts sojourners and permanent settlers alike. Ethnic groups with origins in many countries reside in Modesto including Mexicans, Nicaraguans, Cambodians, Hmong, Vietnamese, Japanese, Indians, Armenians and Indo-Fijians. Indo-Fijians are densely concentrated in Modesto, California, and evidence of the community can be found through various shops, temples, restaurants and associations scattered throughout the city. In Modesto, the size of the foreign-born population from Fiji is estimated at 8 percent (City-Data 2015) while the total foreignborn population from Fiji in the United States is estimated at 39,235 (U.S. Census Bureau: American Community Survey 2013). The majority of these immigrants are of Indo-Fijian descent. Nonetheless, by focusing on the lives of Indo-Fijian women in

Modesto, California, we can begin to understand the distinct gendered experiences of Indo-Fijian immigrants in a specific locale.

Selection of Participants

Before recruiting participants for the study, I obtained approval to conduct research from San Jose State's Institutional Review Board (IRB). This phase consisted of generating flyers for the study, an interview protocol, sample interview questions and informed consent forms. Using a purposive sampling technique for the exploratory study, I began the recruitment process by handing out flyers to potential participants, and friends and families of potential participants. The purposive sampling strategy was useful since Indo-Fijians are a "hidden" population in California: they are not only small in numbers but are often mistaken for subcontinent Indians. Nevertheless, through constant communication with members of the Indo-Fijian community in Modesto, I recruited five Indo-Fijian immigrant women for the study. In order to protect their identities and to keep them from harms way, the women appear in the study under the pseudonyms Krishna, Malti, Priya, Asha, and Sheela (Anthropological Association 2012).

Recruiting women for the study was not as easy task as I thought it would be.

Although I reached out to many Indo-Fijian immigrant women in Modesto, many declined to participate in the study because they believed they would not be of any help to me. Other women showed interest in the study but declined to partake because of work and family obligations. One potential participant cancelled minutes before our meeting. She explained that several years ago her husband had been deported and that

her daughter advised her to not participate in the study, fearing that her participation might lead to deportation.

Description of the Research Participants

This section briefly introduces the five Indo-Fijian immigrant women who participated in the study. Important to note is that the women come from a narrow age group. At the time of the study, the women ranged from the ages of 44-57. Although they are a non-representative sample, I decided to include the women in the study for two reasons. First, as a preliminary exploration into the lives of Indo-Fijian immigrant women in the United States, these participants were willing to share their experiences and perspectives. As mentioned, recruiting women for the study was not an easy task. Although I described the reasons why women declined to participate, I came to realize that some were not willing to partake in the study because they were simply engrained with the idea of being silent as they were in Fiji. Thus, it becomes extremely salient to share the stories of Indo-Fijian immigrant women who were willing to speak about their lives. Secondly, my decision to include the women from a narrow age range is to highlight the variability amongst a group of people of similar ages. Through the silhouettes, we learn that the women may have been of similar ages but their experiences and perspectives are distinctive and personal.

Krishna

Krishna who was 53 at the time of the interviews, emigrated from Fiji to California in 1980. She was the first woman recruited for the study. I met Krishna through the help of an Indo-Fijian friend who had recommended her. I was told that

Krishna had migrated to California as a single mother and would be a good candidate for the study. I gave my friend a flyer to give to Krishna, and after a couple of days, I received a text message from Krishna herself, expressing her interest in the study. After calling Krishna back and explaining to her my research endeavors, she enthusiastically agreed to participate in the study. Interviews, informal conversations and participant-observations occurred at her home. Learning about her experiences and perspectives did not occur in just one sitting, but over several sittings and many hours. The interviews took place on weekday evenings after she arrived home from work. But prior to interviewing her, Krishna always insisted that I have a meal with her and her family. I never declined and it was a pleasure to eat Krishna's home cooked meals. After dinner, Krishna and I would retreat from the rest of her family to her private veranda for our conversations.

Malti

The second woman recruited for the study was 55-year-old Malti, who emigrated from Fiji to Modesto in 1994 with her husband and three children. I was introduced to Malti through a close family member who recommended her for the study. After speaking to Malti via telephone, she agreed to participate. However, Malti was not able to meet at her house for an interview because she spent most of her time working at her family business. Instead, Malti asked for the interview to take place at her work, which was different from the other women who chose to have the interviews in their homes. Nonetheless, I was thrilled to meet Malti at her workplace because I was able to observe her day-to-day activities and tasks there. Although I only had one formal interview with

Malti, our conversation spanned for nearly eight hours. Not only was I able to interview her but I was able to conduct participant-observations at her workplace. After the interview, Malti and I engaged in a series of informal conversations.

Priya

The third woman to participate in the study was 57-year-old Priya whom I had also met through a close family member. After giving a flyer to my relative to give to Priya, she contacted me via telephone to learn about the objectives of the study. I explained to her in detail the research endeavors, and she agreed to participate in the study. Interviews, informal conversations and participant-observations occurred at Priya's home. Priya was extremely welcoming and made chai for us to sip during the interviews. The interviews occurred in two sittings and ended promptly after her grandchildren arrived home from school. With the formal interview ending, participant-observations and informal conversations continued until she would prepare dinner for her family.

Asha

Asha, who was 44 at the time of the interview, migrated to California in 1983 as a 12-year-old adolescent girl. Asha was recommended for the study through an Indo-Fijian friend who gave Asha a flyer on my behalf. After reviewing the flyer, Asha reached me via e-mail. I communicated the goals of the research to her, and she agreed to participate in the study. Without hesitation, she invited me to her home for an interview. Asha greeted me with a delicious cup of homemade chai and samosas prior to the interview. The interview with Asha was effortless; however, her husband was home during the

interview in the next room watching television. Thus, to be respectful to Asha and her husband, I refrained from asking her too many personal questions such as details about her marriage or her role as a stepmother. After the interview was over, Asha gave me permission to call or e-mail her if I had any more questions for her. I took advantage of this opportunity and was able to ask her questions I was not able to during the interview. *Sheela*

Sheela who was 51 at the time of the interview, migrated to California in 1969 at the young age of six. I had met Sheela years ago through a close Indo-Fijian friend, and I reached out to her for an interview. Although I had not seen Sheela for years, she warmly agreed to an interview. Her daughter was also pursuing her master's, and she wanted to help with the study in whatever ways she could. Nevertheless, Sheela and I met during the weekend for a long interview at her home. Prior to the interview, Sheela offered me cookies and coffee, which I happily accepted. The interview with Sheela was different from the other women in the study because due to her early age of arrival in California, her memories of Fiji were sparse. Moreover, the interview with Sheela was the most difficult because her husband was present for half of the conversation. For instance, Sheela often looked to her husband to help answer interview questions. When this happened, it was not always easy to re-direct the conversation back to her. So, to remedy the situation, Sheela and I spoke "off the record" during informal conversations.

Methods and Instruments for Data Collection

Participant Observations

A mainstay in anthropological and ethnographic studies, participant observations were collected for the study and recorded in a field notebook. Participant observations, which can be defined as "the process of learning through exposure to or involvement in the day-to-day or routine activities of the participants in the research setting" (Schensul, et al 1999:91), were used to gain a deeper understanding about the women's lives. By conducting the majority of the participant-observations in the women's homes, I was able to observe the women in their familiar settings. For example, while at Priya and Asha's homes, I noticed the women and their families watching Zee TV, an Indian cable and satellite channel on their televisions. These observations led me to consider the contributions that technology plays in helping the women and their families connect to their ancestral land India, and the ways in which technology creates transnational spaces for the Indo-Fijian community in Modesto.

Furthermore, since what people *do* is often different from what they *say*, participant observations helped to strengthen the reliability and validity of the study. For example, all the women talked about cooking Indo-Fijian cuisine in their interviews. By conducting participant observations, I was able to confirm that the women *do* cook and hold the recipes for traditional Indo-Fijian cuisine. I watched both Krishna and Priya prepare intricate Indo-Fijian foods and drinks such as spicy fish curry and lemongrass *cha* (tea). Moreover, the women also validated their lives and experiences in Fiji by showing me various forms of evidence such as pictures, books, clothing and cookware

that they had brought with them to California. Last but not least, I was also able to confirm their memories, perspectives and experiences with members of their families, or by member checking. This method was useful and helped to develop a more "holistic understanding of the phenomena under study" (Dewalt and Dewalt 2002:92).

In-depth Interviews

The bulk of the data was collected through semi-structured, open-ended in-depth interviews. According to sociologist Shulamit Reinharz (1992), interviews are especially useful in feminist research and allows for accessing the voices of those who are marginalized in a society. She states:

Interviewing offers researchers access to people's ideas, thoughts, and memories in their own words rather than in the words of the researcher. This asset is particularly important for the study of women because in this way learning from women is an antidote to centuries of ignoring women's ideas altogether or having men speak for women. (1992:19)

Thus, interviews were chosen as a data collection instrument because they helped to fulfill the goal of understanding the women's lives from their perspective. Added to this, interviews were also utilized because as Weiss (1994) writes in the introductory paragraph of *Learning From Strangers*:

Interviewing can inform us about the nature of social life. We can learn about the work of occupations and how people fashion careers, about cultures and the values they sponsor, and about the challenges people confront as they live their lives. We can also through interviewing about people's interior experience . . . We can learn the meanings to them of their relationships, their families, their work, and their selves. We can learn about all the experiences, from joy through grief, that together constitute the human condition. (1994:1)

Using this insight, I constructed a specific interview guide, consisting of questions I planned to cover in the interviews. The questions were also open-ended, so the women

could give in-depth factual descriptions about their lives. This allowed for the women to speak in long monologues, and also allowed for capturing the women's emotions, something a "yes" or "no" questionnaire or survey cannot do. Additionally, the semi-structured design of the interviews allowed for probing specific areas of interests. Last but not least, although all the women were fluent speakers of Fiji Hindi, all the women chose to speak English for the interviews; every once in awhile, the women used Fiji Hindi words, phrases or expressions while we conversed. With the women's permission, the interviews were tape-recorded to make sure data was accurately transcribed and not distorted.

Informal Conversations

Some of the most interesting data was attained "off the record," through informal conversations. This was a useful approach and allowed for collecting information not addressed in the interviews. For example, Sheela and I engaged in informal conversations about Miss America 2014, which she brought up on her own accord. Sheela commented on the xenophobic and racist comments made against Nina Davuluri, the first Indian-American to win the title of Miss America. Although Sheela felt upset about the negative comments made toward Davuluri, she was proud that a woman of Indian descent had won the competition. Sheela exclaimed that Davuluri had broken the traditional American standards of beauty that she felt she never fit into. These types of informal conversations helped to enhance the quality of the data by providing an opportunity to engage in natural dialogue and to collect information not easily attained by the semi-structured interviews.

Reflexive Approach

Reflexivity, a tool highly valuable to anthropologists and feminist researchers alike, was utilized for the study. In anthropology, reflexivity is defined as the act of reflecting or "turning back on oneself, a self-reference" (Davies 1999:4). Reflexive anthropology is grounded on the basis that fieldwork is considered an intersubjective process which is created through many subjectivities (Briggs 1970; Geertz 1971; Rabinow 1977), including those of the researchers and their theoretical perspectives, and the perspectives and representations of the study participants (Srinivas 1966, 1979). Reflexivity was an important approach to the study and gave me the opportunity to evaluate my own biases as the researcher. For instance, my identity as a secondgeneration Indo-Fijian woman from Modesto helped me to conceptualize the research. More specifically, being born and raised in Modesto for 18 years of my life by two Indo-Fijian immigrant parents, I have been imbedded in the Indo-Fijian diaspora in Modesto since birth. Not only do I know many Indo-Fijians and speak Fiji Hindi fluently, I also know where and how to access Indo-Fijian "ethnic raw materials" or ethnically linked symbols and practices in Modesto (Jimenez 2010). For instance, I have been shopping at Indo-Fijian markets in Modesto since I was young, consuming and buying staple Indo-Fijian goods such as turmeric, durian fruit, cassaya, incense, booja (spicy snack mix), Bollywood films and Indian clothing. To this day, I still shop at the same Indo-Fijian markets that I went to as a child. I even visit the Indo-Fijian Hindu temples in Modesto and have attended countless pujas, weddings and Hindu festivals in the city such as Diwali and Ram Navami. I also attend Modesto's Fiji Festival, a spinoff of Hayward,

California Fiji Festival, both of which are held annually in the months of June and July, and provide a space for Indo-Fijians to celebrate and experience their unique identity through foods, music, dances, art, and a variety of cultural performances. Thus, my identity as a second-generation woman of the Indo-Fijian diaspora in Modesto allowed me to readily identify the community and gain access into the lives of Indo-Fijian immigrant women.

However, although being an "insider" (Kanuha 2000) of the community had many advantages such as gaining easier access and acceptance to the community and having an understanding of various cultural nuances, it comes with its own set of disadvantages. Perhaps the greatest difficulty of being an insider was balancing the dual role of being an "insider" and a "researcher." For example, while conducting an interview with Sheela, I assumed she was aware of the fact Indians were brought to Fiji as indentured laborers. To my surprise, she did not know this information and in shock exclaimed to me, "My dad never told us we were slaves!" Interestingly, Sheela's husband who was present at the time of the interview, confirmed that Sheela's descendants were girmitiyas. He had discussed her ancestral roots with Sheela's father before he had passed away. After that experience, I began distancing myself from the research, realizing that not all Indo-Fijian immigrants would be privy to Indo-Fijian history. Another moment (which I mention earlier), where I took serious consideration of my role as a researcher was when one potential participant cancelled our meeting because her daughter did not want her mother to face the possibility of deportation like her father had. Although I assured the potential participant that her identity and personal information would be protected, she was not

willing to take the risk. This experience made we aware of the fact that some women might view me as possible legal authority that could affect their lives in negative ways. Nevertheless, these were only some of the ways I affected the research, and I acknowledge that there may be others that are still invisible to me. For example, it is not uncommon for insiders to take something for granted that stands out to outsiders.

Approaches to Data Analysis and Presentation

Transcription

Data analysis involved several steps and was an iterative process. The first step to data analysis was transcribing field notes from participant observations, interviews and interview field notes. To reduce the time-consuming process of data transcription, I used the computer transcription software Express Scribe, which shortens the duration of transcription and enhances the quality with its special software features. For example, Express Scribe features variable speed playback (constant pitch), which allowed me to slow down the speech from the audio-recordings. This feature was especially useful when the women spoke too quickly, or if I had difficulties deciphering through heavy accents.

Thematic Analysis

The next step in analyzing the data set was conducting thematic analysis, which served as a tool for finding meaningful patterns across the data set. This in turn helped form the basis of the anthropological silhouettes. Like transcription, thematic analysis was also an iterative process and occurred throughout the research process. According to R.E Boyatzis, thematic analysis is the process of "encoding qualitative information"

(Boyatzis 1998:vii), or simply the act of developing codes, words or phrase that function as labels for portions of data. After identifying themes, I developed codes. I identified themes by searching for repetition, transitions, indigenous typologies, metaphors and linguistic connectors throughout the data set (Bernard and Ryan 2010). For example, one of the themes found in the research was "significant social networks," and codes and for this theme included "strong ties" and "weak ties," while some of the sub-codes included "family," "friends," and "co-workers." Nonetheless, these types of codes were applied to the entire data set, which was written in the margins of the transcripts.

After coding the data, I began the process of "cutting and pasting," which involved cutting and pasting the codes into piles of codes (Bernard and Ryan 2010:94-95). I was able to do this on my word processor. By this point, I was able to extract data from its original context so I could look for patterns across the data set. For example, while coding the theme "citizenship," I learned that every woman in the study became an American citizen because of the benefits that came along with having citizenship. I also learned that citizenship allowed the women to feel a sense of belonging in the United States. Nevertheless, these types of patterns found in the data set helped to form the basis of the anthropological silhouettes as well as the study's discussion.

Anthropological Silhouettes

In order for the women's voices to be heard, the data is presented in the next chapter as anthropological silhouettes. An anthropological silhouette is a form of lifewriting (the recording of events and experiences of a life), which is considered "less

complete than a biography, and partial, but demonstrably based on an individual and is honest about its limitations and incompleteness" (Zeitlyn 2008:159); like any biography, it is a product of work by the interviewee and interviewer. This approach was not only appropriate for the study because it allows for the women's voices to be heard, but also because the approach recognizes the limitations of documenting people's lives. Moreover, anthropological silhouettes are useful for archiving the lives of people whose lives may not have been documented otherwise. The anthropological silhouettes weave together my voice as the researcher with that of the women to describe their journeys with migration. The silhouettes were assembled by extracting data that provided information on the women's lives as they experienced it in Fiji and their eventual migration and transition to life in California. Furthermore, I assembled the silhouettes by including themes found in and across the data set. Nevertheless, to make the anthropological silhouettes easier to read, I started in chronological order describing the women's lives in Fiji to their eventual migration and transition to life in California. Last but not least, the anthropological silhouettes were also utilized because I could showcase the women as individuals with a set of diverse experiences and perspectives from one another, all while revealing the commonalities that arise amongst their experiences as Indo-Fijian immigrant women residing in Modesto, California.

Limitations of the Methodology

As with all methodologies, quantitative or qualitative, limitations are inevitable. First, although a qualitative approach was used to explore the women's lives, a great deal of data was gathered through participant observations, in-depth interviews and informal

conversations, which made sorting through the large volume of data a time-consuming process. Another limitation to the study included focusing exclusively on women's lives. Although the goal of the research was to learn about Indo-Fijian immigrant women, to gain a more holistic understanding about Indo-Fijian immigrants in California, Indo-Fijian immigrant men must also come under investigation. Nevertheless, despite some of the limitations that I encountered during the study, the goal of the study remains to give a voice to a vulnerable group of people. Marjorie Shostak (1989) describes it best in *Interpreting Women's Lives: Feminist Theory and Personal Narratives:*

It is for Shmuel, !Nisa, and the silent others they represent, as well as for ourselves that we should continue to record these lives and memories. The ethical and methodological problems may be formidable, but they are small compared to the goal. Indeed, the most important ethical message regarding life-histories is not a restriction but an obligation to make every effort to overcome obstacles, to go out and record the memories of people whose ways of life often are preserved only in those memories. And we should do it, urgently, before they disappear. (1989:239)

Although different from life-histories, the same can be said in regards to anthropological silhouettes. Anthropologists have a responsibility of over-coming any challenges so that we can continue to document the lived experiences of people, especially those that have been pushed to the margins.

Chapter Summary

This chapter presents the methodology used to investigate the lives of five Indo-Fijian immigrant women residing in Modesto, California. First, the chapter discussed the framework that guided the study. Next, the chapter provided a discussion on the research design and detailed the selection of participants, methods for data collection and data analysis. Finally, the chapter reviewed some of the limitations of the methodology. The following chapter presents the five anthropological silhouettes, which provide a glimpse into the lives of Krishna, Malti, Priya, Asha, and Sheela. The silhouettes showcase the women's individual experiences with their journey from Fiji to California, which are diverse, complex, and multi-layered. However, after reading each of the silhouettes, it becomes clear that the women share commonalties amongst their experiences as Indo-Fijian immigrant women residing in Modesto, California.

Chapter 4: Anthropological Silhouettes

Introduction

This chapter presents five anthropological silhouettes on the lives of Indo-Fijian immigrant women who appear under the pseudonyms Krishna, Malti, Priya, Asha, and Sheela. As described in the previous chapter, anthropological silhouettes are a form of life-writing, which is considered "less complete than a biography, and partial, but demonstrably based on an individual and is honest about its limitations and incompleteness" (Zeitlyn 2008:159). The anthropological silhouettes were appropriate for the study because the approach allows for the women's voices to be heard.

Nevertheless, the anthropological silhouettes weave together my voice as the researcher with that of the women to present their lives as they once lived them in Fiji, to their eventual migration and transition to life in Modesto, California.

While reading the five anthropological silhouettes, it is important to reflect on the historical context provided in chapter 2. By keeping in mind the history of Indo-Fijians in Fiji, it becomes clear that these women's lives are situated and entangled in a larger social, cultural, political and economic history, which inevitably shaped their lives and precipitated their migration.

Anthropological Silhouette: Krishna

Growing Up in Nausori, Fiji

Krishna was born in 1960 and grew up in the town of Nausori located alongside Fiji's longest, widest and most important waterways: the Rewa River. Raised in a small house with her mother, father, three younger sisters, and a younger brother, Krishna

51

described growing up in poverty:

We were *really* poor. Because I was the oldest one, so I *really* know . . . My dad was the only one who was working and he was only bringing two dollars a week . . . We didn't have enough clothes to wear. My mom used to make clothes for us out of her *saris* (Indian wrap dress). We never wore store bought clothes . . . we didn't even know what a store bought underwear looked like because we never had one.

With no running pipes or electricity, the Rewa River became vital to their daily lives and was used for a variety of everyday tasks:

We use to go wash our clothes over there. We did our dishes over there. We use to take our baths and showers in the river . . . we use to drink the river water . . . and get fish from the river . . . We use to do this because we didn't have any water.

But the Rewa River was not only vital to Krishna's family's daily lives, but also to the daily lives of her Indo-Fijian neighbors. Krishna had fond memories of these neighbors. She reminisced about the friendliness of the Indo-Fijian community and recalled:

Everybody! The whole community would get together . . . that is how it was. Everybody watched out for each other . . . And people use to have animals . . . If somebody in the neighborhood knew that, oh that cow doesn't have water . . . [then] they would get water and they would give it to the cow. That is how it was. It was like that. I miss all that.

Schooling Year: Primary School

Krishna described her primary school years as filled with moments of joy and frustration. The challenging aspects of these years were due to the fact that Krishna was the eldest and also a daughter. In addition to going to school, she was required to fulfill a variety of household tasks. By the age of ten, Krishna's morning routine started by waking up before five o'clock to cook a breakfast of *roti* [bread], curry and *chai* [tea]. She then put together parcels for each member of her family to take for the day. After

this chore, Krishna also helped her siblings dress in their uniform before she quickly prepared herself for the day. But, her responsibilities did not end with the morning routine. Immediately after school, Krishna would go to the river with her siblings to wash their uniforms. After completing these tasks, she would then start her homework until it was time for her to prepare dinner for her family. While Krishna took care of these tasks, her mother spent the day working in the nearby fields:

It was *my* responsibility to help. It was given to me because I already knew *how* to cook. She had chores to do too. She used to go in the field and plant beans and sugar cane and rice and stuff like that. My mom never relaxed for a day. She had a full-time responsibility in the fields . . . she had to go plant and harvest everything . . . that was food for us.

Although *her* responsibility, Krishna often became overwhelmed with everything she had to do. Moreover, school came fraught with its own set of responsibilities. Krishna's school was strict, and girls were required to abide by a series of rigid rules. She was required to dress in a crisp-clean uniform consisting of a blouse and past knee-length skirt, with her hair set in two neat braids. She was also forbidden from wearing makeup or nail polish. Krishna recalled on a friend who was suspended from school for wearing nail polish. She was then reminded of her own personal experience with having to face the consequences of breaking school rules. Krishna described the punishment she received after plagiarizing her friend's math homework:

I was so [bad] in math that I didn't want to do it. I used to get homework everyday. Maybe out of only like 20 questions [math problems], I would do one or two . . . I use to have this friend, [who was in the] same class . . . [and] he used to let me copy his homework. One day I was trying to copy [the assignment] . . . and I didn't fully copy the paper and my teacher had seen me. I was caught and he hit me, then he let me go and he sent me for one hour to go pull weeds. When the one-hour was over, he sent one of the students to get me. I came inside, then he made me go sit crisscross in

front of the classroom. I sat there until end of the period and there was no recess for me . . . it made my self-esteem go down.

Thoroughly humiliated by the experience, Krishna never plagiarized again. Besides this incident, Krishna enjoyed school and her favorite subjects were history and literature. She found Indo-Fijian history especially interesting since her great-great grandparents were *girmitiyas*. According to Krishna, her ancestors were coerced by the British into coming to Fiji and that "they [British] never let them go. They [great-great grandparents] were just left there [Fiji]." Although her ancestors had suffered immensely, she believed the community had progressed since the indentured period.

Secondary School and Abuse

For Krishna, secondary school years were filled with many obstacles, none of which concerned her academics. Her main focus was trying to cope with an abusive, alcoholic father. Krishna explained that her father was extremely strict and threatened her by saying things such as:

'You can't talk to strangers. You can't talk to boys. You can't have boyfriends. You make sure you focus on school and get your education. And if you ever have a boyfriend,' he was going to get me . . . [my father] said he would hit me. He said he would kill me and he would put me in the river.

Scared of her father, Krishna obeyed his rules without question. She did not want to be on his ruthless side since she felt her father was capable of anything when under the influence of alcohol. Krishna recalled a disturbing memory where she had witnessed her father purposefully push her mother off a high platform that broke her leg. Until this very day, Krishna describes being haunted by the traumatic memories of her abusive father. It was not long before Krishna became the target of his abuse. She recalled the

first time she was beaten by her father:

My dad had a girlfriend, and that is one of the other reasons why my dad use to beat my mom. He had this girlfriend on the side . . . Around 17 years old my dad said, 'You know what, we need to marry her.' My mom [disagreed], '[Krishna] is going to school . . . she is young.' My dad said 'No!' Then my mom said, 'Only if you stop seeing that lady! And then my dad said, 'No I can't do that.' [Later, I found out that my father's girlfriend] had an older son . . . My dad told my mom that I'm going to marry her to this boy . . . My mom said 'No! I'm not going to let that happen!' So then my dad beat my mom so much that day that she had a broken jaw. She was bleeding. She was bruised and everything, you know? Since that day, [when] my mom said that to my dad, 'I'm not getting my daughter married!' . . . then, oh my God! Everyday Friday and Saturday my dad would drink . . . and then he would beat my mom, and then he would beat me. He use beat me so bad just because of that!

Infidelity was the catalyst of the on-going abuse in Krishna's household. She described yet another disturbing memory of being brutally attacked by her father:

My sister said [to my dad] 'Krishna won't marry that guy! And then that Friday, my dad came from work. He brought alcohol bottles and put them away. [Then he] told my sister to go cut two or three *dandas* (pieces of wood, sticks or branches) . . . from the hibiscus trees . . . She did it because if she *didn't* do it, she would get hit [by my father]. That is how it was. Most of the time he [would] beat my mom and me up when it got dark. When it [was] night. That night he called me [over to him] and started beating me. He hit me so much, so much, so much . . . with the *danda*! It broke into pieces. Then he picked up another *danda* and then he slapped me. He punched me. [As] he did this I just fell [to] the ground.

For Krishna, this was not the end of the abuse. She continued:

No one knew. So that was the first day. Then every week he used to beat my mom and me. Just because I said I won't marry that boy. My mom [would] cry during the day [when] my dad would be gone at work. My mom would go up to God and pray to God 'Oh God, please save my daughter's life!'

Wishing for her mother's prayers to be heard was not enough to end the violence. One day Krishna was so severely beaten by her father that a piece of the *danda* skewered her

leg. Worried about her daughter's leg, Krishna's mother took her to the hospital without her father's knowledge. When asked by the doctor how her daughter had been injured, Krishna's mother lied about the incident. According to Krishna, her mother had no choice but to lie because at the time the Fijian police did not treat domestic violence cases seriously:

My mom didn't say I got hit. My mom said we were playing on the tree outside and fell and got bruised and the *danda* hit and got in . . . She lied because the police would come to the house and talk to my dad and things would just get worse at the house. They would just warn him and that's about it.

Unfortunately, a brief visit from the police with a warning could potentially worsen the situation at home. The possibility of facing severe retribution from her husband kept Krishna's mother silent about the abuse. Like her mother, Krishna also stayed quiet. *Misfortune of the Bride*

Violence continued in Krishna's household, and her mother believed that the only way to protect Krishna was to have her married. Krishna, who was only 17 and had no desire to marry, and had recently found a job, had no other option but to quit. She explained:

I [had] graduated from that school and applied for a job . . . And I got hired! I went to work [for] one day, the first day, and that was it . . . that was my first day and my last day because the talk of my marriage came around!

Krishna was quickly married to her Indo-Fijian groom. She described the events of her arranged marriage in the following passage:

So then they [in-laws and future husband] came to the house and I was still inside the back room. So they were in the living room talking about it [the marriage]. So they agreed to it. So they set this date . . . I was scared

to death . . . I didn't know the *meaning* of marriage . . . So they agreed . . . Four days later, we had a court marriage . . . And then two weeks later, my real wedding was set up . . . Indian style . . . So then I got married. I was very, very sad . . . depressed. I would cry. Hide myself, and cry . . . I never cried in front of my mom because my mom was always abused and cried everyday. My dad would have killed me if I didn't marry that boy.

Taking the chance, an arranged marriage seemed better to Krishna than the possibility of ending up dead in the river.

Unfortunately for Krishna, the disadvantages of an arranged marriage were all too real. She described the night of her wedding as a horrifying experience. After arriving at her in-laws' house, Krishna's mother-in-law directed her to her new bedroom, where her new groom met her. Krishna, who was not aware of the concept of sex at the time, explained that her new husband raped her. Terrified, Krishna voiced the incident to her mother-in-law who assured her that it was "normal" behavior for a married couple.

Less than a week of being married, tension began brewing at the house. In an explosive fight between her husband and in-laws, Krishna discovered that her new husband never wanted to marry her:

He told me, 'I didn't want to get married! I didn't want to get married! I have a girlfriend!' And he said, 'I wanted to get married to this girlfriend, but she is not my race! She is not from my culture!' She was a different race. She was Samoan. So his parents didn't agree. So they actually lied to him and told him that if you get married to an Indian girl, then we will give you a motorcycle.

Krishna's in-laws never gave their son a motorcycle. Caught in the middle of the conflict, there was little she could do about the situation. For the next several years, Krishna remained a silent bystander in a "loveless" marriage. Cast aside by her husband, she remained under the watchful eye of her mother-in-law, who had quickly come to despise

her:

So, everyday . . . this man [my first husband], he really got pissed. So he said to his mother, 'You take this lady [Krishna], you take this woman [Krishna], keep her with you and I'm going!' So for a few days he left and he stayed with his girlfriend. I was sad. I was crying. I wouldn't eat or anything . . . You know what, then that lady [mother in-law] was mean to me! I guess because her son left and that her son was mad and agitated. So now they were taking everything out on me . . . Then they made me a *slave* there.

From dawn to dusk, Krishna was required to take care of all domestic chores. She also had limited rights in the household and tolerated a plethora of house rules. She explained that her mother-in-law had strict control over the females in the house. Krishna then provided an example of how her sister-in-law endured horrific physical torture for breaking her mother-in-law's rules:

No girls were allowed to go outside of the doors . . . That was the rule from the parents [in-laws]. The girls were not even allowed to look through the window either. If they did, if they catch you, they gonna burn you! One of their daughters', she was 14 years old . . . and it just so happened that she peeked through the window and she said 'Oh *fuwa* (father's sister) is coming!' She said she had seen her *fuwa* coming and her mom heard what her daughter had said, and OH MY GOD! I was there at that time because I was washing [my in-law's] clothes. Her mom goes to the stove and turns it on. Then she took a metal spatula [and] put [the spatula] on the stove [to make] it hot! She made it hot and she took the girl's hand and held it like this (Krishna raised up her palm) and she put that thing there (pointing to her forearm)! I screamed [because] I was scared! She burned her arm from here to here! The skin came off with the metal! Nobody helped her. No doctor. No nothing. That was very evil!

Krishna took the inhumane punishment of her then sister-in-law as a warning. If her mother-in-law could hurt her own daughter, then surely she was surely capable of hurting Krishna. Thus, just as with her father, she obeyed her mother-in-law's rules and remained silent about the violence.

Things were further complicated when Krishna became pregnant with her first child:

One day he comes in the house and he was nice all of a sudden. He was really nice. That day he slept with me. We had sex. And then he was nice. And I'm still emotional. I'm crying and all that. The next day he went to work. He slept with me again. We had sex. And all of a sudden I think I got pregnant! I didn't know what pregnancy was or how you get pregnant! I was really sick. I couldn't keep anything in me and I was feeling weak. So finally one day, he [my husband at the time] seen me . . . that I was going through hell. Half a day he took off. He came. He said okay, you get ready and I'll take you to the doctor. We went to the doctor. This was a native doctor . . . He spoke English and Hindi. And so he wanted to check my urine. So I gave him that. The results came and he said you are pregnant. He told me that! And I said, 'What?' He said 'tumma phet meh bucha' (you have a child in your stomach). And I didn't know what he meant by that! I know phet is stomach, but I didn't know what the bucha (child) was . . . So he told me that you are going to have a baby . . . so I got pregnant. Oh my God, then here came the drama!

Unaware of pregnancy or even how it occurred, Krishna was shocked by the news that she would soon be a mother. The only thing more shocking for Krishna was the ill treatment she received from her mother-in-law. Krishna described her mother-in-law's attempts to abort her daughter:

By then I got pregnant and oh my god, things were not really that good and his mom used to beat me. She tried to give me stuff to drink so I would miscarry. Cause she said you are not supposed to do this. You are not supposed to have this. But nothing worked though. I don't know why. She gave me stuff and it made me sick and throw up, but nothing worked. Nothing worked. . . She knew her son never wanted to get married. He wanted somebody else . . . So that happened and then Radika (her daughter) was born. So all this time I was pregnant, I did not enjoy myself. I did not enjoy my pregnancy. Did not enjoy a thing, *a bit* of my life then because he [my husband] wasn't there. I was all alone. *No family, no money, no friends*. You can't even get out of the house. You can't even talk. No phone, no nothing. No TV. No nothing. Like a *slave*.

Krishna was very serious when she explained that she did not enjoy "a bit" of her life or

her pregnancy. Sadly, due to the drudgery she was living in, she attempted to commit suicide:

All I wanted to do when they kept me inside the house and locked up was commit suicide. Kill myself. Kill myself, but I didn't know how to kill. One day, I decided to hang myself. But I didn't know how . . . So, I pulled the table in the middle to the room. And then, I found a piece of material [from my sari] . . . Then I put a chair on top of the table and I climbed on top of the chair, on top of the table. Then I tied the [sari] on the ceiling and then I didn't know what I was suppose to tie first . . . Then all of a sudden I hear banging on the door . . . [My mother-in-law] started banging [on the door], so I ran to open the door. I got so scared, nervous, I was shaking! I peed in my pants at that time! But, I just ran to open the door. She started screaming! She asked me, 'Why did you take so long to open the door? What were you doing?' And she grabbed me by the hair; she always grabbed my hair to beat me.

At the time of Radika's birth, Krishna's mother, father and siblings had migrated to California through a family sponsorship. But since Krishna was legally married, she could not move to California. She explained:

Two months after I got married, my parents and sisters and brother migrated to America . . . *Make sure you write that down*. Two months after I got married. Because you know what? They had all the paperwork going on . . . My paperwork was going on also with my mom and them so that I could go to America too, but in the meantime my marriage was arranged. It was because of this that I couldn't join my family. That is why I was left out there [Fiji].

With no immediate family support, Krishna remained with her negligent husband and suffered from severe depression.

Abandonment

The turning point of Krishna's marriage was when her husband abandoned her. She described a series of events that eventually led her to divorce:

[We] were still living with Radika's dad but it was just me and Radika living there and he would be gone. And then one day he decided he was

going to take me to my aunt's house. He said, 'I'm going to take you tonight and drop you there and then tomorrow morning I'm going to come and I'm going to bring the baby. So, he kept Radika and took her to his mom and dad's house . . . So then the next day, he never showed up . . . Kept on waiting and waiting, my aunt and uncle and everyone were just waiting. He *never* showed up.

With her newborn child taken from her, Krishna was extremely distraught. Her husband never came back to her, and tracking him down was a major ordeal. After filing a police report, Krishna waited two difficult weeks to learn about her husband and daughter's whereabouts. She detailed the incident:

So you know how we have *Modesto Bee* [Modesto, California's local newspaper] here . . . so in Fiji, it was the Fiji newspaper. So my uncle brought that in the morning. Then I stood at the edge of the dining table and I was just looking at the paper quickly. *Quickly* because I had a lot of work to do. And as I flipped the paper, then I saw his [my husband's] picture . . . I saw his picture and I was like 'WHAT?' . . . And then I started reading. So here's his picture [on the front page] and it had this paragraph about him. His dad put that down in there and it said that he went to India for further studies for five years and that he missed that opportunity to say goodbye to some of his best friends. And so, this is how much I read. I didn't read the whole thing. As soon as I read this part that he went to India for five years for further studies, oh my God, I passed out! I fainted and I fell on the floor!

After returning to consciousness, Krishna was surrounded by her aunt, uncle and cousins. She then proceeded to show them the newspaper. Frustrated by the turn of events, Krishna's uncle darted to the police station with the newspaper in hand. After explaining the situation to the police, she and her uncle were accompanied by law enforcement to her in-law's house, where Radika was then retrieved from a locked room. Krishna was finally reunited with her daughter.

After the chaotic event with her estranged husband, Krishna quickly filed for divorce. For the next two years, she and her daughter lived with her aunt and uncle. This

was a difficult time for Krishna who was unemployed and a single mother of a young daughter. Her relatives could only house the two for so long, and Krishna had to figure out a way to support herself and her daughter. However, once Krishna's parents found out about their daughter's divorce, they quickly sponsored Krishna and Radika to come to the United States. She recalled on the day she received two plane tickets in the mail that changed her life forever. She explained:

Then I get this package that my parents sent from here. I opened it and I seen tickets. Ticket for me and Radika to fly . . . I showed it to my aunt and uncle and they read it and they were all excited. But, I wasn't excited at all . . . I didn't know what was happening to my life at that time, you know? And then my uncle said, 'No this is real, you're going to America!' . . . And within three days of getting that mail, I had to leave.

Coming to America and Entering the American Work Force

Within three days of receiving the package, Krishna and Radika embarked on their long journey from Fiji to California where they were reunited with Krishna's parents and siblings. For the next several years, they lived at her parent's house in Modesto, California. For Krishna, arriving at Modesto was a complete culture shock. Not only was it her first real encounter with "white people," she was shocked by the way people dressed. She remembered feeling embarrassed at the sight of women in bikinis and explained it took her "a good 3 to 4 months to adjust" to American life.

After getting together her legal documents, Krishna began looking for employment in Modesto. She applied for several jobs and took a position as a housekeeper at a care home for elderly. While Krishna worked, her mother and siblings helped to look after Radika. Even though she was only paid \$3.25 an hour for her labor, Krishna was thrilled to earn a paycheck. This was her first step in financially supporting

herself and her daughter. Eventually, Krishna was given more responsibilities at her work until she was promoted from housekeeping to directly assisting clients. The change in her work responsibility was a positive one for Krishna, and she became passionate about her new role as a caregiver. She explained:

I enjoyed caregiving work more than the housekeeping I was doing. And then I got too close to these elderly people. I got so close to them that it was everything for me . . . It was a good thing to take care of other people.

Krishna continued caregiving until sustaining a back injury at work. She has now settled for "lighter" work as a secretary but expressed that caring for the elderly was an incredibly rewarding experience.

Second Arranged Marriage

While living with her parents, Krishna began witnessing her father's belligerent behavior all over again. Although she had hoped her father had changed his abusive ways in the United States, he had not. One evening, in a drunken rage Krishna's father demanded she leave with Radika and never return. The drunken rages continued on for the next month until Krishna's mother finally suggested that her daughter remarry before things worsened. Once again, with the help of her mother, Krishna experienced yet another arranged marriage. Although the memory and experience of her first marriage made the anticipation of another nerve-wrecking, Krishna agreed to her mother's advice. After a few months, Krishna's mother found a perfect Indo-Fijian groom named Raju for her daughter.

Even though Krishna experienced two arranged marriages, the second time around was different, and for the better. Krishna reminisced about her wedding:

The priest came and he did all the things that happen in our [Hindu] custom . . . And then my husband bought me so many nice things and I was just shocked! He opened the bag in front of the priest and everybody and he started giving me all this stuff. He put a ring on me. And, this is all real! He gave me everything real! Real gold bracelet and nothing cheap. He gave me skinny white bangles, bracelets, real earrings, real necklace . . . expensive outfits, *saris* [Indian wrap dress], and all that! Oh my god, I was shocked! All my life everything had been going wrong, you know? Nothing good had happened in my life that I know of . . . I never had a happy day . . . I cried everyday, all my life. I cried, I cried, I cried. That was how my life was.

For Krishna, meeting Raju was a positive, life-changing experience. Not only did he treat her well, he was also kind to Radika. Krishna explained that with Raju she finally felt like a "human-being" rather than a "slave." She believes her husband's kindness and affection toward her stems from his personal experience with an abusive father. Krishna explained that her husband encourages her to stand up for herself. "He says you have to stand up! You cannot be quiet like you used to live! You need to stand up!" Raju helps Krishna to not be held back by her gender. Also, living in California and becoming a citizen of the United States has given Krishna a sense of empowerment and strength. She revealed, "America has made me one strong person, that now, I know that I have a life. I have a life and I can talk. I can stand up for myself, and I can change things." For Krishna, moving to California has changed her life for the better, and for this reason, she does not plan to return to Fiji. She exclaimed, "I don't mind visiting Fiji once and awhile, but I will never live there again. All my family is here now, and I enjoy my rights here, in America."

Purchasing a Home and Raising Children in California

Besides being empowered by her status as a citizen of the United States, Krishna

described feeling empowered when she and her husband purchased their first home.

After residing in an apartment in Modesto for the first few months of being married,

Krishna's and Raju's combined incomes allowed them to make the purchase. Krishna recalled on this moment with excitement:

We started looking for a house and at that time houses were cheap, and the rules and the regulations were not as strict. My income was there, so I got qualified along with my husband . . . So we bought our first house! It was a two-bedroom house, one car garage!

With the comforts of owning their very own home, Krishna and Raju decided to expand their family. Krishna gave birth to their son, Johnny.

Krishna described that both she and her husband are extremely proud of their Indo-Fijian heritage, and passing down their cultural traditions to their children is extremely important to them. Perhaps the most important marker of the Indo-Fijian identity, Fiji Hindi, was passed down to both her children. Radika was already accustomed to Fiji Hindi, which she learned while living with Krishna's parents and siblings upon arrival. Johnny, on the other hand, learned Fiji Hindi from his babysitter, an elderly Indo-Fijian woman whom Krishna had hired while she and her husband worked. From her, Johnny was able to fluently learn Fiji Hindi. Krishna even confessed that her son speaks better Fiji Hindi than she does and that he had learned a great deal from his Indo-Fijian babysitter. Presently, Krishna's children not only connect to their Indo-Fijian roots through Fiji Hindi, they also connected to the Indo-Fijian identity through food. Krishna shared that she taught both of her children how to cook Indo-Fijian foods. Her son, especially, loves to cook. Krishna explained, "Johnny always craves Indian food so he makes it all the time. He loves the food. He makes very good

chicken curry and lamb curry! I am proud of both of my children for knowing about their culture. I hope someday I will have grandchildren and that I can pass on our culture to them."

Anthropological Silhouette: Malti

Growing Up in Nadi

Malti was born on the Western side of Viti Levu, in the quiet countryside of Nadi. She was raised in a large family with three brothers and seven sisters. Being one of the youngest of her siblings, Malti frequently dodged household chores and found ample time to play. She described experiencing a mostly carefree childhood in Fiji where her main responsibility was focusing on school. But since her family was large, money was often scarce in their household. Malti's father, who worked as a carpenter, had difficulties making ends meet while her mother did what she could as a housewife-supplementing the family's survival by growing vegetables and selling them to markets. *Schooling Years*

Unlike her mother and father, and even some of her siblings, Malti was afforded the opportunity of going to school. She explained that she was able to attend school because she had become friends with the headmaster's daughter. Seeing Malti's potential in academics, the headmaster himself paid for her school fees. Everyday, Malti walked three miles to school and excelled in her academics. She proudly remarked, "I was a smart kid. When I was in third grade, I got an award and skipped fourth grade, and was put straight into fifth grade." Her favorite subjects included Hindi and science, and she was proud that she had learned to read and write in Hindi. Unfortunately for Malti, her

schooling was cut short due to financial hardships. She explained "I graduated form 4, but I could not go to school anymore because my dad wasn't working and we did not have any more support."

While out of school, Malti stayed home with her mother and helped her attend to household chores. Under her mother's guidance, Malti became highly skilled in the art of managing a household. Before long, marriage was on the horizon for the young woman. Malti explained that her marriage was unique in that it was a "love marriage" and not an arranged marriage. She added that "love" was key to a long-lasting marriage. At the age of 19, Malti's brother introduced her to an Indo-Fijian man named Vijay living in Suva. After a year of courtship, they were married, and Malti moved to Suva to join her husband.

Moving to Suva, Children, and Remembering the First Coup

Starting a new life in Suva was a shocking yet invigorating experience for Malti. In comparison to rural Nadi, urban Suva was crowded and busy. While living in Suva, Malti gave birth to her three children: two sons and a daughter. Instead of just homemaking for her husband, Malti also acquired the responsibility of motherhood. For the next several years, Malti focused on taking care of her children, while Vijay worked in downtown Suva in a local shop. After working at the shop for a few years, Malti's husband decided to start a family business. With financial help from his father, Vijay rented a stall in downtown Suva and began selling take-out Chinese food.

But shortly after starting their business, Fiji experienced its first coup and downtown Suva was ransacked. Luckily for Malti and her family, their stall escaped

damages, unlike other Indo-Fijian businesses in the area. Malti attributed this to the fact that their business sold "Chinese" food, instead of "Indian" food. She candidly explained the racialized tension that generated the coup:

It [the coup] was because of the Indians and because of the politics [at that time]. Because Indians were more educated [than indigenous Fijians] and Indians were gaining more political power . . . and they [indigenous Fijians] didn't like it.

She also recollected on a terrifying event that she attended with her small children during the 1987 coup:

When the first coup happened, I was in Sukuna Park. We [Indo-Fijians and Fijians] were all gathered together because they were going to release some politicians that had been held over. They were going to release them and have a celebration . . . and I was with my kids and my sister! We all went together, and then we saw big tall [indigenous Fijian] men! They started beating [with their fists] whoever came in front of the . . . And people, out of fear, started running! And when we saw that, we just held our kids and we ran!

But this was only the beginning of the coup. Malti described high levels of panic and chaos in Fiji. She recollected:

And the army came in . . . the Fiji army got control of everything and they put a curfew. No one was supposed to go outside. Nobody was working. Everything was shut down and we never went outside . . . if any person went outside they [Fijians] would snatch your purse! They [Fijians] would snatch your necklace!

Malti continued with her testimony of the events of the 1987 coups. She recalled on houses being pillaged, and a horrific incident involving her Indo-Fijian neighbor:

Three houses away from ours, there was this lady who was raped by three [indigenous Fijian] men . . . She was raped and it was so sad . . . an Indian woman . . . she was all alone at home and she cried and yelled and finally people came to her rescue, but they [indigenous Fijian men] were gone. They put a cloth in her mouth. They tied her up. She had bruises on her face . . . everyone was so scared to go anywhere. It was *very* scary.

This incident instilled fear in Malti and she became very guarded with her daughters.

They were never to be outside alone, and they were no longer allowed to walk to school by themselves.

Eventually, tensions subdued and people returned to their regular routines. But even during the first coup, returning "back to normalcy" was only a myth (Trkna 2008). Fear penetrated the lives of the Indo-Fijian community. Although Vijay re-opened for business, Malti was frightened for herself and her family. After experiencing the chaos in Sukuna Park and learning of her neighbor's rape, Malti was ready to migrate to the United States. However, her family had already been sponsored in 1981 by her sister-in-law, and Malti and her family would have to patiently await their entry to the United States.

From Suva City to "Fairyland"

Managing to survive the dark periods of the coups, Malti and her family waited 12 years for the approval of their paperwork. In 1994, Malti, her husband and three children would board an airplane for the first time. They made the long journey to San Francisco where they were then greeted by Vijay's parents and taken to their house in Modesto. Malti described her first night in California:

We came; it was night . . . there were so many lights outside. *So* many lights out! It was like we were in some *fairyland* (laughs) . . . Fiji was always dark . . . and then we moved to the house my mother in-law and father in-law bought for us. And they were staying there already so we moved there at night with all of our bags and we had the bedroom there, and the bed was set! Everything was ready for us! *Everything* was ready! I had my two sons, so they shared a room together. And, my daughter, she was six years old, and she shared a bed with my sister in-law.

With the help of family, the move from Fiji to California was made a little easier. What was difficult, however, was finding employment. With her limited work experience in Fiji, Malti remained jobless in Modesto for two years. As for Vijay, finding a decent job was just as challenging. Feeling like he had no other options, Vijay settled for a job as an ice cream truck driver, which he despised. Furthermore, his wages were inadequate to support the entire family. This situation led Vijay to once again ask his father for financial help with starting a brand new family business. Vijay's father agreed, and together the couple started a new business selling Indian goods. But, starting a business in California had its own set of challenges. Malti sacrificed spending quality time with her children, who took a backseat to the everyday demands of American life. This was a difficult transition for Malti who prior to migrating had dedicated most of her time to raising her children.

Raising Children in California

When Malti arrived to California, her children were all under the age of ten, and one of the main reasons for emigrating from Fiji was to offer them a better life. Malti felt that the United States would allow her children to have better opportunities for education and jobs. This in turn would enhance their quality of life. Thus, she encouraged her children to go to college. Although Malti's eldest son decided not to go to college, he decided to work full-time for the family's business, a decision Malti fully supports. Malti's other's son is a graduate of ITT Technical Institute and is now a successful computer technician. Her daughter attended Phoenix University and also helps with the

family business. Malti's is proud of all three of her children and explained that they all have strong work ethics.

Malti's children are still very connected to their Indo-Fijian identity, despite emigrating as youths. All three of her children are fluent speakers of Fiji Hindi and are aware of their ancestor's indentured past. Additionally, Malti passed on the art of Indo-Fijian cooking to her daughter, who she raves is an exceptional cook like herself. Besides having less quality time with her children while they were younger, raising children in California has been a positive experience for Malti. She is now focusing her efforts on passing down aspects of the Indo-Fijian culture to her grandchildren such Fiji Hindi and food recipes.

Converting to Christianity

Both Malti and her husband grew up in religious Hindu households and wanted to continue their religious traditions for their children. But, after arriving to California, Malti learned about Christianity and connected to the faith. She explained:

I'm going to say, you know, about how I came to the Lord . . . My mother-in-law, father-in-law, sister-in-law- they were all Christian. They use to go to church. So slowly . . . I started to go to church . . . I really liked it. I started myself in the choir . . . and I fully gave my life to the Lord. And I gave myself to the Lord and accepted Jesus as a savior of my life.

Although the church that Malti first attended in Modesto was primarily white, the church she now attends caters to the Indo-Fijian and Indian community. She explained, "We have an Indian community church on Sundays . . . this Sunday I'm talking . . . I preach in Hindi once a month. And my pastor preaches in both Hindi and English." Malti even

described owning a Hindi Bible, which she ordered from India. She explained how her congregation empowers her:

You know, I was *very* shy. As I grew up, I was a very shy person. I would hardly talk. I would hardly talk to people. When I came to the Lord, he gave me the opportunity to open myself up, just like how I opened myself up to you and now I can share my life. I can share whatever I know with other people.

Not only is Malti more extroverted since finding her new faith and supportive congregation, she also attributes her family's business success and the purchasing of their home to converting to Christianity: "When I gave my life to the Lord, we got our business. And when we got this business we bought a house!"

Proud to Be an American

In 2004, Malti and her entire family applied to become naturalized citizens of the United States. After studying meticulously for their exams, the family became citizens of the United States. Malti was ecstatic in gaining the rights and privileges of her adopted homeland. She described:

It was tough to memorize all those things . . . you had to remember several things . . . it was good, though. It [citizenship] has its benefits. It is good because when you have citizenship, then you can have some place where you can stand and have a *place* . . . you know you are American . . . Just like America accepted me, I have to accept America . . . I am an American citizen!

Additionally, Malti described what "being American," meant to her:

When you are American, you have freedom . . . You can live your life. *You can move forward*. You can take better steps in your life and not get stuck in the past. *Just move forward* . . . and by doing that for yourself, you can help others achieve a better life.

Migrating to California has allowed Malti and her family to "move forward" and "achieve a better life" just as they hoped for.

Visiting Fiji

Since migrating, Malti makes frequent trips back to Viti Levu to visit her relatives that still reside there. She also makes visits to Fiji for her family's business. For Malti, Fiji is still a true paradise despite its political instability and "hatred" toward Indo-Fijians. She explained, "Sometimes I go to Fiji, just for 10 days to relax." Malti now considers Fiji especially relaxed in comparison to life in California. She explained:

Fiji life is relaxed. Even though you don't work that much . . . But here in America, you have to work hard, struggle, and then you have to pay so many bills, and then sometimes you don't see your husband or your children when you have to work different shifts . . . that is why we go to Fiji . . . just to relax.

Although Malti loves her country of birth, she has no plans of returning to Fiji. She explained that even though Fiji is in a calm state, a coup could potentially erupt at any time. Added to this, Malti was also fearful of tropical storms that could arrive without a moment's notice. Fiji is located in an area that is highly prone to the occurrences of tropical cyclones. In fact, since 1840, 153 tropical storms have been recorded in Fiji (Chung 1987) and Malti's parents have been victims of such storms. Malti recalled on a storm that nearly claimed their lives. She described:

You know, my husband says all the time, 'When I retire, I will buy a house there [in Fiji]' and says we will go there. And I say, 'No! I'm scared of hurricanes.' Too many hurricanes in Fiji and they would only come by one part of the island, where we lived, called the Western side. There were lots of storms in Nadi. When I was married, my mom and dad went through this hurricane, when [my dad] was sick and paralyzed. He was in bed when the hurricane came [and] they didn't want to move. The roof flew away. Then part of the roof went down and [my mother] put my

dad under the bed. When she was just getting out from under the bed one of the beams fell on my mom's head and [on] her back . . . she had a deep cut on her head [and] had a lot of bleeding . . . but she saved my dad!

Although her parents survived the natural disaster, Malti still fears potential storms.

Anthropological Silhouette: Priya

Early Life in Fiji

Priya was born in 1953 in rural Nausori and lived nearby the Rewa River. Like many other Indo-Fijians in her community, Priya grew up in poverty. Her father, a farmer without education, had difficulties providing for his wife and four children.

Poverty had kept Priya from receiving a secondary education; however, she was proud to have at the very least a primary level education, where she learned to read and write in Hindi and English.

As the youngest and only daughter in her family, Priya described receiving considerable affection from her parents. Priya spent much of her early childhood with her maternal grandmother, a former *girmitiya*, who had lived to be over 100 years old. She often told Priya stories about her life. She recalled on her memories with her grandmother:

She told me stories about India. She was not even married in India but they [maternal grandparents] came in the same boat and they got married in Fiji. My *nani* (maternal grandmother) was really sad when she came to Fiji. They had no life. They were missing their life in India, like their childhood. So they came [to Fiji] and had to work the whole day and they had been whipped by the supervisors . . . the British people. They worked under the British . . . My grandma worked on the sugarcane farm . . . Life was very hard for them. They [British] made them *slaves*. They worked hard on the sugarcane farm.

Considering the tone of Priya's voice, it was apparent that life as a *girmitiya* was full of pain and sorrow. Although the *girmitiyas* had suffered tremendously, Priya acknowledged their resiliency. "They worked hard and they made a little bit of money, but they survived."

Marriage, Family, and Tensions from the Coup

Priya's teenage years were spent preparing to become a wife. Her mother taught her how to cook, clean and manage a household. When Priya turned 18 years old, she married a handsome Indo-Fijian man, and the young bride moved across the island from Nausori to the western side of Ba to join him. It was there that Priya gave birth to her children, two daughters. Priya referred to motherhood as her most important role in Fiji, where her main focus was looking after her daughters and making sure they received a good education.

During the coups, Priya felt tension but for the most part, felt safe in the outskirts of Ba. For Priya, the coup was generated by racialized politics. She exclaimed:

They [indigenous Fijians] did not want them [Indians] to become the big bosses, president, vice president . . . They [native Fijians] said they owned Fiji and that it was *their* country . . . but I don't think so because people from *India* came and built it. When they came, they [Indians] had to work with sugar and then they [Indians] used their minds to build a house, a school . . . they [Indians] cleaned all of Fiji. It used to be a jungle . . . so they cleaned everything, worked in the fields, and made money. They helped to make Fiji.

For Priya, the *girmitiyas* such as her grandmother had made major contributions to Fiji, however, with the expiry of their land leases, Priya and her husband were ready to leave Fiji. Also, with the coups, there were no guarantees for their futures. Thus, emigrating seemed a better option than risking their livelihoods in Fiji. As a result of the coups,

Priya asked her brother to sponsor them to California. Without hesitation, he quickly filed their paperwork.

Moving to Modesto

In 1994, after seven years of waiting to come to California, Priya and her entire family arrived to the San Francisco International Airport where she was reunited with her brother after nine long years. Ironically though, it was a bitter experience for her two daughters, who after departing realized that they could no longer spend time with their *aji* (paternal grandmother) and *aja* (paternal grandmother). But the thought of her daughters remaining in Fiji was distressing for Priya and she mentioned the difficulties they could have possibly faced if they remained in Fiji:

[At the time] Fiji didn't have many rights for ladies . . . they [women] just went to school, and then after they could not find jobs and so they stayed home . . . men would get hired even though women would be more educated, but they would hire men first . . . and over their they [Indo-Fijian women] get married at 18, 19.

These were real fears Priya had for her daughters. She wanted something more than a domestic life for them, and the move to California would offer her daughters the opportunities to become independent women. Currently, both of Priya's daughters have obtained associate degrees and are working in the medical field.

Employment and Buying a House

Staying with her brother was only a temporary living arrangement until Priya and her husband could find employment. Both quickly found work in low-wage paying jobs. Priya explained, "I did housekeeping. I worked in a motel and then I worked in a hospital [housekeeping] I worked 11 years here [in the United States]. Work in the hospital

wasn't so hard . . . but motel work was harder because they gave us too many rooms with not enough employees." As for Priya's husband, moving to the United States brought challenges to his career as an accountant. If he wanted to continue on as an accountant in the United States, he would have to repeat his education and training. But, with a wife and two daughters to take care of, Priya's husband found a job as a kitchen helper at a local hospital instead. After a few months of working at the hospital, he applied for a new job with a corporation and was hired as an agent. Eventually, he worked his way up to a managerial position where he started to make a decent living.

After only six months of living with her brother's family, Priya and her family moved into an apartment and then their first home, "We got the jobs and then we moved into the apartment . . . we stayed there from 1994 to 1996 and then in 1996 we bought this house." This was a huge success for Priya and her husband because she described that Indo-Fijians like herself had no land in Fiji and that being able to purchase their own home provided them with a sense of security and establishment in California. Priya described, "In Fiji, since the land did not belong to us, but to the Fijians, there was no sense to build a house because as soon as the leases would expire, then we would have to breakdown our house because it would still be on their [indigenous Fijian landowners] property."

The Return to Fiji

Since migrating to California, Priya has visited Fiji three times. She explained that after arriving to the United States, her husband vowed to never return to Fiji. Priya explained:

When he came here he was happy . . . because he struggled all his life. When he was really young, his mom died. Then my father-in-law got remarried, so then these kids were harassed by a new, young stepmother. She was not nice when they were young . . . He said 'I'm not going back!' but he went twice to visit!

Even a ruthless stepmother could not keep Priya's husband away from his native homeland, his father or siblings. As for Priya, visiting Fiji is enough and she has no desire to move back. She exclaimed, "There are more rights here, in this country [United States of America] than in Fiji." These rights provide Priya and her family the consistency they desired, which was the most convincing factor in their decision to remain in California.

Citizenship: Becoming American

Knowing that she wanted to permanently reside in the United States, Priya, along with her husband and two daughters, became naturalized citizen of the United States in 2001. Priya described the advantages of their new status and exclaimed, "The benefit of becoming a citizen, [it means] that you can travel anywhere with your passport without a visa . . . and then you can sponsor your family . . . and that you can vote." Although these were some of the answers that Priya had rehearsed for her exam, becoming a citizen also meant being able to offer her family a better life. She expressed "It means a better life for my kids, my grandkids . . . they don't have to suffer the way we did when we were their age! So, they don't have to suffer like that. Like, we didn't have the things that they have *now* when we were their age."

A Visit to the Motherland India

Priya and her husband began traveling the world together and consider this luxury as an opportunity she gained after arriving to California. With two incomes to pay for travel, the couple visited New Zealand, Australia, Canada and Singapore. Priya also visited India after being invited by family friends whose relatives reside in New Delhi. Priya decided to take the opportunity and made the long 17-hour flight from California to the motherland India. Although excited to visit, Priya described being overwhelmed by the country of her ancestral roots. She explained, "India is very well now. But India is very crowded . . . very crowded! There are a lot of prayer places and lots of religious things . . . The temples are very crowded that they can't see the statues. People are also very poor in India. It is very sad." Priya offered a comparison between India, Fiji and the United States of America, explaining, "Fiji is better than India, but America is better than both Fiji and India . . . I am very happy that I live in America because my life is better than most of the people who live in Fiji and India."

Raising Grandchildren

Priya has two grandchildren from her eldest daughter. As a grandmother, Priya has taken the responsibility of teaching her grandchildren about their Indo-Fijian background and culture. Priya explained:

I am teaching Hindi to my grandkids now . . . they understand, but they can only speak a little . . . They know how to say *roti* (bread), *thercari* (curry), *bhindi* (okra), *bhaji* (leafy greens) . . . they like *gulab jamun* (dessert) and *ladoos* (dessert) . . . they like Indian clothes . . . They dance and sing from watching Zee TV.

Through teaching her grandchildren Fiji Hindi, exposing them to the array of Indian foods, and introducing them to Bollywood culture through satellite television, Priya plays an active role in transferring the Indo-Fijian identity to them. Furthermore, Priya's grandchildren's natural curiosity and pride about their identity has helped in preserving aspects of the Indo-Fijian identity in their family.

Anthropological Silhouette: Asha

Growing Up in Suva, Fiji

Asha began the story of her life in Fiji by first describing the death of her father when she was only three years old. Although she had no particular memories of her father, she recalled the difficulties of his passing for her mother and seven siblings. The loss of Asha's father not only caused emotional distress on her family but also severe financial restraints. As a housewife and newly single parent, Asha's mother devised a plan to earn an income. Asha explained:

My dad was an engineer. He built roads and he had his own business until he passed away. But after he died, we didn't have much money . . . We ended up selling all the tractors and everything. Then [with the money] my mom built a two story so she could rent it out.

Her mother's cleverness helped to sustain the family for a while. Furthermore, to offset the cost of food, Asha and her siblings spent much of their time working on their brother-in-law's farm in an exchange for vegetables and grains. Sometimes Asha would go to the farmer's market with her brother-in-law to sell the vegetables. She described:

In Fiji, my brother-in-law would take his vegetables, every Friday and Saturday to the market and sell it. I loved going to the market with him! I would go over to his school right after school on Friday. He lived very far away. I had to take two buses. I was only nine years old.

Added to her delightful memories of the farmer markets, Asha also described enjoying the company of her friendly Indo-Fijian neighbors:

When I was growing up . . . you just walked over to the neighbors and you didn't have to say hey, are you going to be home? You just walked up and they welcomed you whenever you went over. And over here you can't just go to the neighbors . . . the neighbors would come over, like Friday nights and Saturday nights . . . They would go to the market on Saturday, so all the neighbors would go to the market and they will bring all the *bhaji* (leafy greens) and bean and whatever and they'll sit in the front porch and help each other clean it, and peel it, and all that, and then there were times when neighbors would come and all the kids would come. And while the women were doing that, the kids would all play. We had different Indian games like *buli danda* [similar to cricket] that we would play together.

Asha enjoyed spending time with the Indo-Fijian community, but her days in Fiji were limited upon her migration to California.

Leaving Home at 12

When Asha was 12 years old, her maternal grandmother suggested she move to California to live with her uncle. Not only would Asha be able to receive an American education, she could also one day sponsor her family to come to the United States. Asha who was far from keen on the idea of leaving Fiji, explained:

I didn't come here because my parents were here. I came because my grandmother lived with my uncle . . . my dad had passed away, so my mom was living with all her eight children and my grandma thought it would be a better opportunity for us all to come. So, when my grandma went back from the United States to Fiji, to visit, she tried to get my sister to come but her visa didn't go through. So, she only had one week left so she said, 'Why don't I try hers [Asha].' So when we applied, in like two days I got my visa. And, then I had to come to the United States. I wasn't happy about coming because I didn't want to leave my mom.

Asha vividly remembered the day she left Fiji. It was a cold night in December of 1983 and she arrived at the San Francisco International Airport, where she was greeted by her

uncle and aunt and taken to their home in South San Francisco. She described feeling tremendously anxious about her departure. She recalled:

I didn't really want to come. I didn't know what the United States was . . . what it was going to look like but I *had* to come. I was *only* 12 years old and I remember coming here and crying *every* single night for about 2 years straight, and I cried and cried and cried because I wanted to go back home and be with my mom.

But Asha explained that she could not return to Fiji because she was given the obligation of sponsoring her family for "better education, better environment, and growth."

Moreover, the possibility of returning to Fiji was further diminished after her adoption.

Although Asha originally arrived to the United States under a student visa, it was her adoption by her uncle that allowed her to remain in the country. Asha explained:

So what my uncle did was he adopted me and I needed to be adopted before I turned 16, because my student visa wasn't going to last that long so in order for me to stay, he needed to adopt me so I could live here. So like two days before 16, he adopted me but in order for my adoption to go through faster, on the paperwork they had said that both my parents were deceased.

Asha's adoption was expedited because of the fabrication. Fearful of the law, neither Asha nor her uncle ever attempted to sponsor the rest of their family. She explained:

So, if I were to apply for my brother and sisters, their thing would have said their mom is alive and mine would have said they are both deceased, so I couldn't apply for them.

Although her siblings never had the opportunity of migrating to the United States, two of Asha's sisters migrated to Vancouver, Canada after the coups, where they now live as permanent residents. Having spent so many years separated from her siblings, Asha visits them often to help make up for lost time.

Growing Up in South San Francisco

Asha spent the next several years of her life living with her uncle, aunt, and cousins in South San Francisco. Immediately after arriving to California, Asha was enrolled at a local high school instead of junior high. As an immigrant, she was instantly enrolled in English as a Second Language (ESL) classes; however, the classes did not apply to her. She explained:

In Fiji, they teach you English. So I knew English but I still had to take the classes but then when I took the classes they were like what (laughs)! So then I had to just take regular English classes.

While Asha excelled in academics, socializing with her peers was difficult. Asha's high school, which was incredibly diverse, included a growing number of Indo-Fijian students. Her efforts to initiate friendships with Indo-Fijian girls were met with hostility. The girls bullied Asha for her fashion choices, but she ignored the mistreatment. She explained:

I don't like to follow what other people do. I have my own personality and I'm not going to change because someone says I need to change . . . I just stopped hanging out with people like that.

Asha's strength and character allowed her to move on and make new friends.

After graduating high school at the age of sixteen, her uncle enrolled her into accounting classes. Seeing Asha's potential for success, he hoped that one day she would become a successful Certified Public Accountant (CPA) like himself. But Asha found accounting tedious. Instead, she applied for a job at a private elementary school where she was offered a position as a teacher's aide. Through this job, Asha decided to dedicate her life to mentoring underprivileged children. To this day, Asha works with underprivileged children and plans to continue doing so to until her retirement.

Marriage and American-Born Children

Since the age of 16, Asha described receiving several marriage proposals from various Indo-Fijian families and suitors. But marriage was not on the young woman's mind. Asha focused her time and energy on her education and career. It was not until Asha turned 21 that she had met her future husband, an Indo-Fijian man named Arjun. The couple fell in love and married, and Asha moved to Modesto to join her husband and new stepdaughter. In the years to follow, Asha and Arjun would have three children together: two daughters and a son. At this point in Asha's life, her main focus was raising her stepdaughter and three children. As a mentor, she instilled the value of education in her children, and, as a result of her influence, all her children are attending major universities in and out of California with their futures looking prosperous.

Furthermore, Asha and her husband have instilled aspects of the Indo-Fijian identity in their children. For example, Asha explained that her children openly embrace their Indo-Fijian roots. Asha's daughters, Chandra and Pinky, are especially connected to their cultural background:

My beliefs, my faith is something that I passed on . . . The *pujas* (Hindu prayers) I would do in the morning, my girls do them now and the mantras and things like that, they do it now. And they like the clothing, [and] cooking. Pinky makes *bhindi* (okras) and *roti* (bread).

Asha's daughters also enjoy Hindi music and continue to share their love for it despite being away at college:

Like the other day, Chandra went online and started sending [emails of] songs to us. When my kids were little I would always listen to the Hindi songs so they would know them. And now she is sending me all these songs that she use to listen to when she was little and then Pinky started it too . . . they are missing home.

Readily having access to Hindi music via the Internet allows Asha's daughters to stay connected to their Indo-Fijian identity.

Returning to the Homeland

Since 1990, Asha has visited Fiji several times, and her most recent visit was in 2010. Asha described that between her first visit back to Fiji in 1990 to her latest visit in 2010, the island had faced major developments. The roads, streets, houses and people that she once knew had all disappeared. Although Fiji is no longer the Fiji she once known, Asha still plans to continue her visit to her native country. However, she has no desire of permanently returning to Fiji. Asha exclaimed, "this [California] is home for me now. I have built a life here and my family is here. So, I don't mind visiting Fiji, but I will never move back there." With no intentions of ever moving back to Fiji, Asha is content with her life in the United States.

Anthropological Silhouette: Sheela

Early Life, Memories of Fiji

Sheela was born in 1963 in Nausori, Viti Levu and grew up in a small house with her mother, father, and four siblings. Being the youngest child in her family, Sheela described her childhood as "laid back" and "carefree." Her best memories of Fiji revolved around her favorite foods. She raved about succulent mangoes, sweet crab and tasty coconuts. Although these foods were abundant in Fiji's natural environment, Sheela grew up in poverty. Sheela's mother, a housewife with no education, and Sheela's father, a sugar cane worker with little education, had major financial difficulties. And to make matter worse for Sheela's father, work was not always available for him in Fiji. When

this was the case, he would go to New Zealand to partake in seasonal low-waged agricultural work. Sheela described that having her father away from home was difficult on her family and that her father hoped to find employment that would keep him closer to home.

The Move to Colusa County

With little opportunities for employment in Fiji, Sheela's father decided to move his family to the United States of America. In 1969, a six-year-old Sheela and her entire family left Fiji to live in a small agricultural town in Colusa County, California.

Sponsored by her uncle (mother's brother), he encouraged the family to migrate to California since opportunities for steady agricultural work were plentiful. But, the move to California was shocking for Sheela who realized she no longer could spend time with her favorite aunt who she had left behind in Fiji. This was a traumatic experience for Sheela not only because she left her aunt behind, it was also the last time she would ever see her aunt who passed away after her emigration from Fiji.

Within months of their arrival, Sheela's uncle, who was also an agricultural worker, helped her entire family find employment in the farming industry. Even though Sheela was only a little girl at the time, she worked side-by-side with her family in the fields. She described:

The work was hard . . . the *way* we worked. We worked really hard with our parents . . . And farm life is hard. It is not easy being in the direct sunlight with no trees.

Although work was strenuous and difficult, the combined income of the household allowed the family to quickly save up money. Eventually, after a year of working in the fields, Sheela's family moved into their own home in Colusa County.

Schooling

Immediately after arriving to California, Sheela was enrolled at a local elementary school. She had never gone to school and did not speak English. She described being terrified for her first day of kindergarten. Learning English was a major challenge for Sheela but luckily her eldest sister, an intermediate English speaker was able to tutor her. As for Sheela's parents, they could not help her with school. Both were poor English speakers. Sheela's mother only knew basic words such as "yes," "no," and "hello," and Sheela's father could hold a conversation in English if necessary. In fact, Sheela's father scorned English and made a rule that while at home, no one was allowed to speak in English. Sheela explained, "My dad said 'I will never speak to you in English!' He *never* did!" By allowing only Fiji Hindi to be spoken in their home, Sheela's father was helping to preserve his family's native language.

The Move to Modesto and High School Years

In 1978, Sheela's family moved from Colusa County to Modesto, California. This decision was intentional. Sheela's eldest sister was recently married to an Indo-Fijian man and moved to Modesto to live with him. After learning about various job opportunities in the Central Valley, Sheela's father moved the rest of the family to Modesto. Sheela was not thrilled with this decision and described feeling "very unhappy" about leaving the countryside. She had grown to love Colusa County and

moving away from her uncle, aunt, and cousins was difficult. She was also shocked to move from a small rural town to a much bigger city. To make matters worse for Sheela, she was also about to enter high school, an experience she described as "horrible." Not only did Sheela describe having difficulties with her academics, especially with English and math, she also described being bullied by a group of girls at her school. By her senior year, she had given up on high school. She believed having a job and money was more important than an education.

Marriage

Sheela's parents discussed the topic of marriage with her when she was still in high school. Sheela exclaimed that her father never encouraged her or her sisters to go to college as he had with her two brothers. Instead, Sheela's father wanted his daughters married. She lamented:

I wish my dad told me, "Hey go to school and get your education.' I wish he was more like that . . . he never did all of that. He wanted me married. He wanted me married at 17.

But Sheela did not marry until she was 19 years old, when her aunt introduced to her to her future husband. After meeting her potential groom once, Sheela was married to her Indo-Fijian husband Ajay, who at the time had recently migrated to California.

Employment

Sheela found her first job during her sophomore year of high school. She was hired as a busser at a Mexican restaurant. Sheela worked at the restaurant for several years before being hired at her current job of over 25 years as a housekeeper at a hospital. While housekeeping, she had applied for other jobs and found employment as a secretary.

But Sheela described that her lack of computer skills resulted in being "let-go" from the job. This was a point of contention for Sheela who described having low self-esteem when it came to computers and technology. However, Sheela exclaimed that she might give computers another chance in the future and that it is never "too late" to go back to school.

Child-Rearing

Sheela and her husband have two daughters, and both are currently living outside of Modesto. Both of Sheela's daughters are California State University graduates, something she and her husband are very proud of. Since Sheela never graduated high school or attended college, she encouraged her daughters to focus on their education. She explained:

I do housekeeping . . . if I *had* the choice, I wouldn't do it! But it's not hard money. I'm comfortable where I am . . . but I would never want my daughters to do that ever!

For Sheela and her husband, raising children in California also means instilling in them values found in both the Indo-Fijian culture and the American culture. For example, Sheela described that she was strict with her daughter's while they were growing up just like her father was with her:

In a way I was glad that he was strict, maybe it kept me from trouble . . . I was strict but in a difference sense. Cause I always laid it on them, 'Look how I was raised!'

Sheela's daughters were aware of the strict Indo-Fijian culture that their mother came from. For instance, Sheela's father refused to speak to Sheela in English; he also refused to speak English to his grandchildren. Sheela explained:

He refused to speak English to us and he told my youngest one, Kayla, 'I will never speak to you in English because you have got to learn your language!' She was weak at speaking the language. Thank God he was strict because I felt Kayla learn more Hindi because of him too.

She recognized that it was her father's way of making sure that the future generations of their family knew something about their cultural roots. For Sheela's daughters, they yearn to be connected to their Indo-Fijian identity and Sheela elaborated that her daughter blames her for not speaking to them in Hindi at home: "She blames me! 'You didn't tell us more or teach us! Oh, you talk English'!"

Converting to Catholicism

Although Sheela grew up in a religious Hindu home, she later converted to Catholicism. Sheela explained that she learned about the Catholic religion from her eldest daughter's best friend. With her, Sheela's daughter had attended many Catholic functions such as volunteering to feed the hungry, and other charity events. Sheela's husband was so impressed with "all the good" that he decided to attend mass and urged Sheela to come as well. With an open mind, Sheela went to mass and enjoyed her experience there. She began regularly attending church and described seeing many positive changes in her and her family's life. For instance, Sheela's husband quit drinking alcohol and is living a much healthier life. Also, Sheela and her husband renewed their vows and are enjoying watching their daughters grow into strong, independent women.

Sponsoring Family During the Coups

Sheela had a limited understanding of Indo-Fijian history. In fact, she never knew that Indians were indentured to Fiji and that her ancestors were *girmitiyas*. She

exclaimed, "My dad never told me we were *slaves*!" Sheela was shocked to find out that Indians were indentured servants in Fiji. Her husband who was present during the conversation confirmed that Sheela's ancestors were indentured. Ajay recalled having talked to Sheela's father about his family's *girmit* past. This then led Sheela to recall on sponsoring family members from Fiji to the United States:

Well, I will be honest . . . I said to him [points to Ajay] 'Hey, why don't you call on at least one of your family members here?' So he sponsored his brother's family . . . When he came we had to run around a lot to immigration trying to explain why they wanted to stay here and they kept asking him why he wanted to stay here and he said the coup was going on in Fiji.

Sheela was happy to have the family at her house and described feeling empathy toward them, especially since they also had two young daughters. She also remembered being sponsored and housed by her own uncle and Sheela was happy to return the favor to her brother-in-law

No Return. Yet

Since migrating to the United States, Sheela has not been back to Fiji. She exclaimed:

I haven't been back because I have a little bit of a phobia . . . the day we got on the plane was scary. There was a lot of turbulence and that day we thought we were going to die.

The fear of crashing that day created a phobia of flying for Sheela, which still remains with her today. But, Sheela's hopes to conquer her fear of flying so she can one day show her daughters the country of her birth.

Chapter Summary

This chapter presents the data collected for the study in the form of anthropological silhouettes, which chronicles the lives of five Indo-Fijian immigrant women residing in Modesto, California. The silhouettes showcased a diverse set of experiences and perspective amongst the women. Although the women's experiences are diverse, it becomes apparent after reading all the silhouettes that common themes arise amongst their experiences as Indo-Fijian immigrant women living in Modesto. The next chapter provides a discussion to the study, which focuses on the major commonalities and themes found amongst the women's experiences. It also describes the implications and contributions of the study.

Chapter 5: Discussion

Introduction

This chapter provides a discussion of the anthropological silhouettes that chronicle the lives of five Indo-Fijian immigrant women residing in Modesto, California. The silhouettes offer a glimpse into the women's lives as they once lived them in Fiji to their eventual migration and transition to life in California. When read individually, the silhouettes showcase a diverse set of experiences amongst the women. They indicate that migration is a personal experience and that the women are not a homogenous group; they are *individuals* with *unique* experiences and perspectives. At the same time, when read collectively, the silhouettes reveal common experiences and perspectives as Indo-Fijian immigrant women. This chapter explores the major commonalities and themes, and concludes with a discussion of the implications and contributions of the research.

Push Factors: Economic and Political

Collectively, the anthropological silhouettes highlight that economic and sociopolitical factors pushed the women's emigration from Fiji. Fathers were identified as the
sole wage-earners in their households, which created severe financial constraints for their
families. Although mothers were typically not wage-earners, their contributions were
vital to their families' livelihoods. Their responsibilities included but were not limited to:
managing the domestic household, child-rearing, working in the fields, and producing
and processing food for consumption. Inadequate finances, in combination with large
families, presented economic hardships that affected the women's lives. Both Malti and
Priya were unable to progress educationally because of the high cost of school fees.

These women had little choice but to stay at home with their mothers and focus on the art of domesticity. Krishna described growing up in poverty and lacking basic necessities such as clothing, water and electricity. Lacking water and electricity was stressful and added extra responsibilities for the then young Krishna, who spent a great deal of her time at the Rewa River washing clothes and dishes, and fetching water to be used at home.

Economic burdens and constraints not only affected the women's lives in Fiji, they directly propelled their emigration. As a divorced, single mother with "no family, no money, no friends," Krishna migrated to Modesto in 1980, where she joined her parents and siblings. With her family's support, Krishna re-built her life. She found employment that allowed her to support her daughter. She eventually purchased a home with her husband. Doing so empowered Krishna, especially given her past in Fiji. Asha also departed from Fiji in 1983 due to economic hardships. She emigrated to alleviate her mother's financial stress caused by raising eight children on her own and to ultimately sponsor her family for "better education, better environment and growth." Finally, Sheela migrated in 1969 with her family because of the lack of job opportunities for her father in Fiji. Although the focus in this thesis has been on women's lives, to effectively assess their decisions to emigrate, family structures and the economic resources available to both men and women must come under investigation.

Comingled with economic reasons for emigrating from Fiji were also sociopolitical motives. As explored in the second chapter, Indo-Fijians have been politically marginalized in Fiji since they first arrived. Indo-Fijians were deterred from participating in Fiji's politics due to racialized sentiment and the fear of Indian domination in the nation's politics. Malti and Priya expressed the sentiment that political instability in Fiji had solidified their desire to depart their country of birth. The coups led both women to feel uncertain about their future lives if they remained in Fiji. Malti was afraid for herself and the safety of daughters, and Priya expressed her worries about the expiration of her husband's land lease. As landless people, land leases were vital to Indo-Fijian survival in Fiji. Without the political right to own land and with landlords refusing to renew existing land leases to their tenants, many Indo-Fijians thought they had no place in Fiji. Thus, the potential risks of migrating to California were offset by the present risks in Fiji while they lived there. Moreover, decisions to emigrate from Fiji were warranted especially after reflecting on Fiji's history. Both Malti and Priya's families emigrated in 1994, which enabled them to escape the 2000 and 2006 coups. Furthermore, Priya and her family departed the islands just three years prior to the 1997 expiry of land leases in Fiji. They were able to avoid the possibility of being internally displaced in Fiji and living in one of Fiji's squatter settlements.

The Nature of Migration: Family Sponsorship

Another theme that emerged from the silhouettes was family-sponsored migration. All the women were sponsored to the United States by relatives already established in California. Each woman had either consanguineal (blood) or affinal (kin by marriage) relatives who sponsored their relocation to the United States. Krishna was sponsored by her parents, Malti by her sister-in-law, Priya by her brother, and both Asha and Sheela were sponsored by their uncles. With the help of these relatives and through

the U.S. Family Reunification Program, the women and their families obtained legal entrance into the United States. The program offers legal entry into the United States off two broad subcategories including immediate relatives of U.S. citizens and family sponsorship according to preference categories (Ramah McKay 2003).

Their kin networks not only helped the women gain legal entry to the United States, but they supported the women and their families as they transitioned to life in California. Relatives provided shelter for the women and their families until they were financially capable of moving into their own households. For instance, Malti's in-laws bought a large house for her family to live in, which was also fully furnished with all the necessities. Moreover, Asha's uncle helped her family find employment and Krishna's siblings offered free childcare for her daughter Radika. These relatives provided vital systems of support for the women and their families. Such support in turn enabled them to experience a relatively smoother transition to life in California than those without established kinship networks in a receiving country.

Reuniting and Separating From Family

Another theme that appeared in the silhouettes was that of family separation and reunification. As previously described, relatives helped the women and their families gain legal entrance into the United States, which was not only to provide them with better social, political and economic opportunities than Fiji could, but *also* to simply reunite with their relatives. For example, Priya who was sponsored by her brother described being overjoyed after reuniting with him after nine long years and Krishna was relieved to reunite with her parents and siblings after being abandoned by her

husband in Fiji. While reuniting with family was a joyous occasion for the women, it was only one side of the migration coin. The other reality that the women faced with their emigration from Fiji was leaving behind members of their nuclear and extended families. For instance, Asha, who emigrated independently at the age of 12, described suffering from years of separation anxiety from her mother and siblings. Similarly, Sheela was only six years old when she was separated from her favorite aunt, who had unfortunately passed away in Fiji before she even had the prospect of reuniting with her. Although these were challenging experiences for Asha and Sheela, they are both resilient and managed to overcome these incidents of separation and loss.

Nevertheless, although the issue of family separation and reunification is found amongst all refugee and immigrant groups, the fact that the women chose to speak about their experiences with reuniting and separating from family, underlines the importance of the topic amongst people who migrate. It points to the fact that migration effects people both psychologically and emotionally, and that even with time, the experiences of migration are not forgotten.

Social, Political, and Economic Mobility

Since migrating to California, the women have collectively experienced upward social, political and economic mobility. Socially, the women had been confined to the private domestic sphere in Fiji, where from a young age they were prepared to become obedient wives and mothers. Since migrating to California the women are no longer confined solely to the domestic sphere and they now must engage in the public sphere. With more freedoms in their movements, the women have acquired opportunities to alter

parts of their identities. For example, both Malti and Sheela were Hindus prior to migrating, had converted to different religions after migrating to California. Malti converted to Christianity after going to church with her in-laws and Sheela described converting to Catholicism after learning of the creed from her daughter's friend and through attending mass.

The women have also increased their political rights since migrating to California. Referring back to the historical context, the political future of Indo-Fijians generally and Indo-Fijian women specifically were considered bleak. As Indo-Fijians, their rights were limited and they were excluded from Fiji's political system. They could not own land and to this day still cannot. As a result, having political rights is important to the women and has led them to become naturalized citizens of the United States of America, which ensures that they are protected under the U.S. Constitution and Bill of Rights. They also gain many rights and benefits through citizenship such as voting, sponsorship of family members to the U.S., eligibility for federal jobs and financial aid, and Supplemental Security Income (SSI). However, the most significant benefit of citizenship is that it offers the women a sense of belonging and the opportunity to permanently reside in California. Krishna, Malti, Priya, Asha, and Sheela have all chosen to reside in the United States of America permanently. With California as their new permanent home, the women have no plans of re-migrating to Fiji. Instead, visits, vacations and memories of their country of birth suffice for the women who have now transitioned into Indo-Fijian-Americans.

Finally, the women have increased their economic power since arriving in California. Krishna, Malti, and Priya were all housewives in Fiji but after migrating they joined the American labor force. This change in their gender role was necessary for adapting to their new lives in California and it allowed the women to begin earning independent incomes. These incomes have given the women the opportunity to purchase homes, send their children to college and technical schools, and have the luxury of travelling the world.

Maintaining the Indo-Fijian Identity and Not Just Immigrants but Transmigrants

None of the women plan to re-migrate to Fiji due to their increased social, political and economic power in the United States, but they stay connected to selected aspects of their Indo-Fijian identities. Jointly, it is important for the women to maintain aspects of the Indo-Fijian identity in California and to pass them onto their children and grandchildren. As fluent speakers of Fiji Hindi and an important marker of Indo-Fijian identity, "Fiji Baat" is passed down to their children and grandchildren. For example, Krishna's children are fluent speakers of Fiji Hindi. Her daughter learned Hindi Fiji while living with Krishna's parents and siblings, and Krishna's son learned the language from his Indo-Fijian babysitter. Krishna explained that hiring an Indo-Fijian babysitter was a useful strategy because her son Johnny learned to speak Fiji Hindi fluently from her. Moreover, Malti and Priya are teaching their grandchildren Fiji Hindi, and Sheela's daughter aspires to fluency in the language. Besides language, the art of Indo-Fijian cuisine is also preserved amongst the women and their families. Most of the women were taught to cook Indo-Fijian cuisine by their mothers during their childhood. These

same recipes are now being passed on to their children and grandchildren regardless of gender, which also suggests ways in which gender roles have been altered since migration.

All the women identified as Hindus prior to migrating, but religious affiliation was not strongly preserved. However, this does not mean that Hinduism is unimportant in the Indo-Fijian community in Modesto. Religion appears to be an important marker of the Indo-Fijian identity in Modesto and especially via the visibility of Indo-Fijian temples, ashrams, and religious organizations in the city. Furthermore, through participant-observations at the religious institutions, it is clear that both Indo-Fijian immigrant men and women are making efforts to preserve Hinduism in Modesto. Nevertheless, further research is needed to understand religious maintenance and conversion amongst members of the Indo-Fijian community.

The silhouettes also suggest that the women are best described as transmigrants and not just immigrants. Transmigrants are defined as:

Immigrants whose daily lives depend on multiple and constant interconnections across international borders and whose public identities are configured in relationship to more than one nation-state. They are not sojourners because they settle and become incorporated in the economy and political institutions, localities, and patterns of daily life in the country in which they reside. However, at the very same time, they are engaged elsewhere in the sense that they maintain connections, build institutions, conduct transactions, and influence the local and national in the countries from which they emigrated. (Schiller, Basch, et al 1995)

Each woman engaged in varying degrees with both Fiji and India. Even Sheela, who migrated at the young age of six and has not been to Fiji since, still maintains ties to her country of birth. She housed relatives during Fiji's coups, continues to shops at Indo-

Fijian and Indian markets, and transmits aspects of her Indo-Fijian identity to her daughters. Also, technology such as satellite television, the Internet, and airplane flights makes it easy for the women to maintain transnational ties to both Fiji and India. For example, to familiarize her grandchildren with their Indian roots, Priya watches Zee TV with them. Asha's daughters use the Internet to listen and share Bollywood music with her. And finally, all the women (besides Sheela due to her phobia of flying) have made visits back to Fiji while Priya has visited India.

Women's Resiliency and Immigration as Success

The anthropological silhouettes demonstrate the resiliency of the Indo-Fijian immigrant women in this study. Each woman described facing adversities and challenges in both Fiji and in California and overcoming them. The sources of their resiliency were diverse. The women found strength in familial sources such as through relatives who sponsored them or the thought of their children suffering in Fiji. Others became resilient through spiritual sources while some found strength through their own personal characteristics. Overall, for these women, their migration can be interpreted as a success story. Through the women's stories we learn that they, along with their families, hoped that departing Fiji would help to resolve issues and conflicts that they hoped to overcome in the United States. For the majority of the women and their families, migration resolved their issues, and as Malti describes, allowed them to "move forward" with their lives. This is especially evident when set against the backdrop of the disturbing history of Indo-Fijians in Fiji, the experiences they encountered there, and the lives they once lived.

Their increased social, political and economic mobility since migrating and their desire to permanently reside in the United States are also symbols of their success.

Implications and Contributions of the Research

In my quest to give voice to Indo-Fijian women and to simultaneously learn about Indo-Fijian emigration to the United States, this study has several implications and hopefully contributes to our knowledge about this population. First, the anthropological silhouettes help to reveal needs of Indo-Fijian immigrant women in the city of Modesto. The needs identified also extend to their families, the Indo-Fijian immigrant community at large, and newly arriving immigrants. The first need identified is the need for emotional and psychological support services. Although the women are resilient and have overcome a variety of obstacles; emotional and psychological support services may help the women better cope with the stresses they once encountered in Fiji and the strains of migrating and transitioning to life in a new country. For instance, while in Fiji, Krishna experienced domestic violence not only from her father but also her mother-inlaw. Depression and anxiety from these experiences as well as a crumbling marriage led to her to attempt to suicide. Although migrating allowed her to move on from such experiences and focus on rebuilding her life, Krishna admitted that I was the *first* person she ever spoke about her life to. She also mentioned that her experiences with her father in Fiji and in California ultimately severed their relationship. Also, while living in Fiji, Malti experienced some unnerving events during the 1987 coups, which had led her to live in fear and worry about the safety of her daughters. Also, as a 12-year-old adolescent girl, Asha's departure from Fiji without her mother and siblings was a traumatic

experience. And, after experiencing heavy turbulence on the airplane from Fiji to California, Sheela has developed a serious phobia of flying that she still carries with her today. Since the women described coping with the issues by themselves, and often in silence without much support, psychological and emotional support services may be beneficial to the women and could potentially aid in their mental health and wellness.

The study also helped to identify the need for vocational and professional services for Indo-Fijian immigrant women *and* men in California. The study revealed that most of the women and their families had difficulties finding employment. With no prior work experience in Fiji, combined with low levels of educational achievement, most of the women were limited to low-wage, unskilled employment in California. According to Berger (2004), making vocational and professional services available to immigrant women can help to empower them:

Training and employment are important parts of the healing process. Employment is powerful leverage for achieving and experiencing success because it allows women to interact with the social environment. The workplace is an important arena for assisting the process of establishing relationships within the wider community and offering opportunities to progress in acquiring the language and practicing it, leading to a sense of worth and power. Therefore, development of employment promotion programs that help women to develop work search, job keeping, gaining and retaining employment, and vocational/professional skills are necessary components of services for immigrant women. (2004:214)

Thus, vocational and professional services can assist Indo-Fijian immigrant women achieve greater employment success. For example, Sheela tried to advance from housekeeping. Although she had found a secretarial job and was on her way to better pay, she was "let go" due to her lack of computer skills. Even though Sheela describes housekeeping as "easy" work, she expressed that she would never want her daughters to

be housekeepers. She aspires for them to become professionals. Also, we learn through the women that men in the community also face challenges with employment. For instance, Priya described that her husband was an accountant in Fiji. Although he had the necessary skills for accounting in the United States, he was required to retrain. But, as a newly arrived immigrant with a wife and two children, Priya's husband had no choice but to find immediate work to support his family. He took a job as a kitchen helper at a hospital. Furthermore, Priya had mentioned that her husband was intimidated by American education and schooling. Nonetheless, professional and vocational services may be especially useful and empowering for recent arrivals who may lack qualifications, or for those who might need retraining and additional education.

Finally, there is a need for community programs for Indo-Fijian immigrant women. Berger (2004) states that community programs should focus on:

Education about legal rights, especially, but not exclusively, for those who come from totalitarian and oppressive societies, in which speaking out was dangerous and could have led to vindictive disciplinary actions, and for those who come from traditional societies that oppress women. (2004:228)

Coming from Fiji where the women and their ancestors have been oppressed and marginalized because of their ethnicity as well as their gender, points to the need for community programs, where the women can be educated about their political, economic and social rights in California and in the United States. Furthermore, the community programs should also offer ways in which the community can preserve their unique cultural heritage. Classes on Fiji Hindi, the history of Indo-Fijians, and Indo-Fijian cuisine can be especially beneficial to the community who is seeking to preserve and

maintain aspects of their Indo-Fijian identity. Although the Indo-Fijian community in Modesto may provide these services informally, they are not easy to identify.

Finally and generally speaking, this study makes a contribution because the women's stories help to put human faces to U.S. immigration. Since immigration is one of the most pressing contemporary social issues in the United States, the ways in which immigrants are depicted, especially in the mainstream, becomes extremely salient. Many misconceptions regarding these immigrants exist in the United States. Sociologist Pierette Hondageneu-Sotelo (1994) describes these misconceptions in her book *Gendered Transitions: Mexican Experiences of Immigration*:

First, immigrants were blamed for stealing jobs of U.S. citizens and depressing wage levels . . . Second immigrants were accused of draining the U.S. economy through their consumption of social services. The accusations included claims that immigrants come to the United States to obtain welfare payments, that they do not pay taxes, and that their children and families constitute a growing underclass, as they drain medical and educational resources in the United States. Finally . . . new immigrants from Asia and Latin America were [perceived] after all "too different," that they were ultimately inassimilable. (1994:xv)

These misconceptions continue to persist as immigrants are rarely given opportunities to defend themselves or voice their experiences and perceptions. By studying the lives of immigrants, listening to their voices and reflecting on the history from which they come from, many of the misconceptions can be debunked. For instance, the history provided in this study, accompanied with the anthropological silhouettes, help to do just this. We learn that the women came to the United States to not steal jobs, but instead to perform jobs that most Americans refuse to do. We learn through exploring their lives that they contribute to the American economy through their participation in the labor force as well

as through buying homes and sending their children to universities and technical schools in the nation. Moreover, we learn that they are *not* "very different" from the native population after all. They too, like their "native" counterparts, have the same hopes and dreams of increasing the quality of their lives and their families' lives.

Finally, although the Indo-Fijian immigrant women's experiences and perspectives are not unique and are widespread amongst all immigrants, their stories are merely a reflections, or in this case silhouettes, of the phenomenon of migration. These reflections call attention to widespread immigrant needs, which is extremely significant in a nation where the foreign born population has hit a record-high of 42.5 million as of July 2014 (Camarota and Zeigler 2015:1).

References Cited

Aggrawal, Ravina

2000 Traversing Lines of Control: Feminist Anthropology Today. The Annals of the American Academy of Political and Social Science. 571: 14-30.

American Anthropological Association

2012 Statement on Ethics: Principles of Professional Responsibilities. Arlington, VA: American Anthropological Association.

Asian Pacific Migration Research Network

Nd Migration Issues in the Pacific: Issues in Fiji. http://www.unesco.org/most/apmrnwp6.htm, accessed April 1, 2014.

Bhabha, Homi K.

1994 The Location of Culture. London: Routledge.

Berger, Roni

2011 Immigrant Women Tell Their Stories. New York: Routledge.

Bernard, Russell and Gery W. Ryan

2010 Analyzing Qualitative Data: Systematic Approaches. Thousand Oaks: Sage Publications.

Boyatziz, R.E.

1998 Transforming Qualitative Information: Thematic Analysis and Code Development. Thousand Oaks: SAGE Publications.

Briggs, Jean

1970 Never in Anger: Portrait of an Eskimo Family. Cambridge: Harvard University Press.

Camarota, Steven and Karen Zeigler

2015 U.S. Immigrant Pop. Hit Record 42.4 Million in 2014: Center for Immigration Studies. http://cis.org/us-immigrant-pop-hit-record-42-million-2014, accessed October 28, 2015.

Chung, J.

1987 Fiji, Land of Tropical Cyclones and Hurricanes: A Case Study of Agricultural Rehabilitation. *Disasters* 11(1): 40–48.

City-Data

2015 95355 Zip Code (Modesto California) Detailed Profile. http://www.city-data.com/zips/95355.html, accessed September 26, 2015.

Cooper, Stonehewer

1879 Fiji: Its Resources and Prospects. London: G. Street and Company.

Davies, Charlotte A.

1999 Reflexive Ethnography: A Guide to Researching Selves and Others. New York: Routledge.

Davis, Mark, prod.

2001 Bitter Harvest. Special Broadcasting Service. Video, 21:30. http://www.sbs.com.au/news/dateline/story/bitter-harvest, accessed December 10, 2014.

Devi, Monica

2012 The Twice-Migrants: Origins of the Indo-Fijian American Identity in the East Bay (CA). MA thesis. Department of History, California State University East Bay.

Dewalt, Kathleen M. and Billie R. Dewalt

2002 Participant-Observation: A Guide For Field Workers. Walnut Creek: AltaMira Press.

Dobell, Graeme

2007 The Strange Saga of Speight's Siege in Suva. *In* Coup: Reflections on the Political Crisis of Fiji. Lal, Brij and Michael Pretes, eds. Pp. 98-102. Canberra: ANU E Press.

Fiji's Women's Rights Movement

2006 Herstory: Celebrating 25 Years of Balancing Scales 1986-2011. Society of International Development 49(1): 1-42.

Fraenkel, Jon

2007 From Election to coup in Fiji: The 2006 Campaign and Its Aftermath. Canberra: ANU E Press.

Geertz, Clifford

1971 Thick Description: Toward an Interpretive Theory of Culture. *In* The Interpretation of Culture. C Geertz, ed. Pp. 3-30. New York: Basic Books.

Gillion, Kenneth L.

1956 The Sources of Fiji Indian Emigration to Fiji. Population Studies 1(2): 139-157.

1962 Fiji Indian Migrants: A History to the End of Indenture in 1920. New York: Oxford University Press.

Gilroy, Paul

1993 Small Acts: Thoughts on the Politics of Black Culture: London: Serpent's Tail.

Gounder, F.

2011 Indentured Identities: Resistance and Accommodation in Plantation-Era Fiji. Amsterdam: John Benjamins.

Hondagneu-Sotelo, Pierrette

2003 Gender and U.S. Immigration: Contemporary Trends. Berkeley: University of Press California.

Hyam, Ronald

1990 Empire and Sexuality: The British Experience. New York: Manchester University Press.

Jimenez, Tomas

2010 Replenished Ethnicity: Mexican Americans, Immigration, and Identity. Berkeley: University of California Press.

Kanuha, Valli Kalei

2000 Being Native Versus Going Native: Conducting Social Work Researcher as an Insider. Social Work 45(2): 439-447.

Kelly, John

1991 A Politics of Virtue: Hinduism, Sexuality, and Counter-Colonial Discourse in Fiji. Chicago: University of Chicago Press.

Lal, Brij

- 1985a Kunti's Cry: Indentured Women on Fiji's Plantations. Indian Economic and Social History Review. 22: 55-71.
- 1985b Veil of Dishonour: Sexual Jealousy and Suicide on Fiji Plantations. Journal of Pacific History 20(3): 135-155.
- 1992 Broken Wave: A History of the Fiji Islands in the Twentieth Century. Honolulu: University of Hawaii Press.
- 1998 Crossing the Kala Pani: A Documentary History of Indian Indenture in Fiji. Canberra: Australian National University.
- 2001 Chalo Jahaji: On a Journey Through Indenture In Fiji. Canberra: Australian National University.
- 2003 Fiji Islands: From Immigration to Emigration. Migration Policy Organization. http://www.migrationpolicy.org/article/fiji-islands-immigration-emigration, accessed April 1, 2014.

- 2004a Bittersweet: The Indo-Fijian Experience. Canberra: Pandanus Book.
- 2004b Girmitiyas: The Origins of Fiji Indians. Lautoka: Fiji Institute of Applied Studies.
- 2010 In the Eye of the Storm: Jai Ram Reddy and the Politics of Postcolonial Fiji. Canberra: ANU E Press.
- 2012 Fiji Before the Storm: Elections and the Politics of Development. Canberra ANU E Press.

Lal, Victor

1990 Fiji: Coups in Paradise: Race, Politics, and Military Intervention. London: Zed Books Ltd.

Lamphere, Louise

1996 Gender. *In* Encyclopedia of Cultural Anthropology, Vol 2. D. and M. Ember, eds. Pp. 448-493. New York: Henry Holt and Company.

Lateef, Shireen

1987 Indo-Fijian Women: Past and Present. Manushi No. 39: 1-10.

Lewin, Ellen

2006 Feminist Anthropology: A Reader. Malden: Blackwell Publishing.

Mangubhai, Francis and France Mugler

2006 The Language Situation in Fiji. In Language Planning and Policy in the Pacific, Volume 1: Fiji, The Philippines, and Vanuatu: New York: Multilingual Matters, Ltd.

Mayer, Adrian

1963 Indians in Fiji. London: Oxford University Press.

McKay, Ramah

2003 Family Reunification. Migration Policy Institute.
http://www.migrationpolicy.org/article/family-reunification, accessed March 14, 2014

Miller, Christopher

2008 A Community of Sentiment: Indo-Fijian Music and Identity Discourse in Fiji. Ph.D. dissertation, Department of Ethnomusicology, University of California Los Angeles.

Mishra, Vijay

2007 Literature of the Indian Diaspora: Theorizing the Diasporic Imaginary. London and New York: Routledge.

Nicole, Robert

2011 Disturbing History: Resistance in Colonial Fiji. Honolulu: University of Hawaii Press.

Premdas, Ralph R.

1995 Ethnic Conflict and Development: The Case of Fiji. Hants: Avebury: Ashgate Publishing Company.

Rabinow, Paul

1997 Reflections of Fieldwork in Morocco. Berkeley: University of California Press.

Radio Australia

2012 Disturbing Indo-Fijian Suicide Rates, radio, 4:12. http://www.radioaustralia.net.au/international/radio/onairhighlights/disturbing-indo-fijian-suicide-rates, accessed April 1, 2014.

Radio New Zealand International

2014 Fiji PM Apologises Over Brain Drain.

http://www.radionz.co.nz/international/pacific-news/281078/fiji-pm-apologises-over-brain-drain, accessed August 12, 2014.

Reinharz, Shulamit

1992 Feminist Methods in Social Research. New York: Oxford University Press.

Revington, Mark

2008 Guns and Money *In* Coup: Reflections on the Political Crisis of Fiji. Lal, Brij and Michael Pretes, eds. Pp. 98-102. Canberra: ANU E Press.

Robie, D.

2001 Coup Coup Land: The Press and the Putsch in Fiji. Asia Pacific Media Educator 10(16): 148-162).

Rabuka, Sitiveni

2012 The Fiji Islands in Transition: Personal Reflection. *In* Fiji Before the Storm. Brij Lal, eds. Pp. 13-20. Canberra: ANU E Press.

Ratuva, Stephen

2007 The Pre-election 'Cold War': The Role of Fiji Military During the 2006 Election. *In* From Election to Coup in Fiji: The 2006 Campaign and Its Aftermath. Canberra: ANU E Press.

Rosaldo, M.Z.

1980 The Use and Abuse of Anthropology: Reflections on Feminism and Cross-Cultural Understanding. Signs 5(3): 389-417.

Rubin, Gayle

1975 The Traffic of Women: Notes of the Political Economy. *In* Towards Anthropology of Women. Rayna R. Reiter, ed. Pp. 157- 210. New York: Monthly Review Press.

Sanadhya, Totaram

1991 [1876] My Twenty-One Years in the Fiji Islands: The Story of the Haunted Lines. John Kelly and Uttra Kumari Singh, trans. Suva: Fiji Museum.

Schiller, Nina Glick, Linda Basch, with Cristina Szanton-Blanc

1995 From Immigrant to Transmigrant: Theorizing Transnational Migration. Anthropological Quarterly 68(1): 48-63.

Schensul, S.L, Jean. J. Schensul, with Margaret D. LeCompte

1999 Essential Ethnographic Methods: Observations, Interviews, and Questionnaires. Walnut Creek. AltaMira Press.

Sharpham, J.

2000 Rabuka of Fiji: The Authorized Biography of Major General Sitiveni Rabuka. Queensland: Queensland University Press.

Shostak, Marjorie

1989 What the Wind Won't Take Away: The Genesis of Nisa- The Life and Words of !Kung Woman. *In* Interpreting Women's Lives: Feminist Theory and Personal Narratives. Personal Narrative Group, eds. Pp. 228-240. Bloomington: Indiana University Press.

Srinivas, M. N., ed.

- 1996 Some Thoughts on the Study of One's Own Society. *In* Social Change in Modern India. Pp. 147-163. Berkeley: University of California.
- 1979 The Fieldworker and the Field: A Village in Karnataka. *In* The Field-worker and the Field: Problems and Challenges in Sociological Investigations, M. Srinivas, A.M. Shaw, E.A. Ramaswamy, eds. Pp. 19-28. Oxford: Oxford University Press.

Stoghill, Ron and Amanda Bower

2002 Welcome to America's Most Diverse City. Time Magazine. http://content.time.com/time/nation/article/0,8599,340694,00.html, accessed June 20, 2015.

Stacey, Judith

1988 Can There Be a Feminist Ethnography? Women's Studies International Forum 11(1): 21-27.

Strathern, Marilyn

1987 An Awkward Relationship: The Case of Feminism and Anthropology. Signs 12(2): 276-292.

Subramani

1979 The Indo-Fijian Experience: St. Lucia: University of Queensland Press.

Tavola, Helen

1992 Secondary Education in Fiji: A Key to the Future. Suva: Quality Print Limited.

Tinker, Hugh

1974 A New System of Slavery: The Export of Indian Labor Overseas, 1830-1920. New York: Oxford University Press.

Trkna, Susanna

2008 State of Suffering: Political Violence and Community Survival in Fiji. Ithaca: Cornell University Press.

U.S. Census Bureau: California Quick Facts

2014 California: People Quick Facts.

http://quickfacts.census.gov/qfd/states/06000.html, accessed December 30, 2014.

U.S. Census Bureau: Modesto Quick Facts.

2014 Modesto, California: People Quick Facts.

http://guickfacts.census.gov/qfd/states/06/0648354.html, accessed April 15, 2015.

U.S. Census Bureau

2013 2009-2013 American Community Survey Five-Year Estimates.

http://factfinder.census.gov/faces/tableservices/jsf/pages/productview.xhtml?pid=ACS 11 5YR B05006&prodType=table, accessed October 30, 2015.

Visweswaran, Kamala

1994 Fictions of Feminist Ethnography. Minneapolis: University of Minnesota Press.

Voigt-Graf, Carmen

- 2001 In Search of a Place to Call Home: The Emergence of an Indo-Fijian Transnational Community. *In* Migration in the Asia Pacific: Population, Settlement and Citizenship Issues. R. Iredale, C. Hawksley, and S. Castles. Pp. 367-387. Sydney: Edward Elgar.
- 2008 Transnationalism and the Indo-Fijian Diaspora: The Relationship of Indo-Fijians and Its People. Journal of Intercultural Studies 29(1): 81-109.
- 2010 Hindu Communities of the Fiji Islands. *In* Religion of the World: A Comprehensive Encyclopedia of Beliefs and Practices. J. Gordon Melton and Martin Baumann, eds. Pp. 1111-1113. Santa Barbara: ABC-CLIO, LLC.

Weiss, R.S.

1994 Learning From Strangers. New York: Free Press.

World Bank

2015 Population, Total: Data Table. http://data.worldbank.org/indicator/SP.POP.TOTL?page=5, accessed October 30, 2015.

Zeitlyn, David

2008 Life-history Writing and the Anthropological Silhouette. Social Anthropology 16(2): 154-171.

Appendix A: Interview Consent Form

Consent Form

Agreement to Participate in Research

Responsible Investigator: Ambrita Nand

Title of Study: Mothers and Daughters of the Indo-Fijian-American Diaspora: First-Generation and Second-Generation Indo-Fijian Women Residing in California's Central Valley

- 1. You have been asked to participate in a research study, which seeks to investigate the lives of women belonging to the Indo-Fijian American diaspora residing in Modesto, CA. This research on the Indo-Fijian women residing in Modesto is being conducted towards the completion of a master's degree in applied anthropology at San Jose State University. The research will be used in a thesis that will be submitted for review by the Department of Anthropology. There is also a possibility that research may also be presented at academic conferences or published within academia. The goal of this research is to understand the gendered experiences of first-generation and second-generation Indo-Fijians women. This research will be used to gain a better understanding on how identity and cultures are transformed and reinvented in a new time and space.
- 2. You will be asked participate in various discussions concerning the migration experience, ways of preserving traditional Indo-Fijian culture, Indo-Fijian history, how the different generations differ, and the changing roles of women in this community. Interviews with specific Indo-Fijian informants will be used in collecting information. These interviews will be vital in understanding women's experiences with migration as well as to understand the experiences of second-generation women living in Modesto, CA. The interviews will take approximately sixty minutes but participants are free to lengthen or shorten their interviews due to comfort. If you agree with participating in the interviews, you will be asked to have the conversations audio recorded. This will be beneficial so that information is not distorted. However, you may refuse to be audio recorded, and in its place, written notes shall be taken. Furthermore, the informants can ask the researcher for personal information to be excluded from the research such as: their names, workplaces, etc.
- 3. There is little or no risk in participating in this research project beyond that of discussing and reflecting upon your own personal experiences. Potential risks may involve a loss of privacy or exposure. Interview material such as the audio recordings, notes, and transcripts will be securely stored in locked filing cabinets. No one besides the researcher will have access to these materials. All personal information will be kept confidential to the extent allowed by law. Agencies with research oversight, such as the SJSU Human Subjects-Institutional Review Board, have the authority to review research records.
- 4. No direct benefits are expected from the research. However, indirect benefits include the general feeling of reward for participating in research where no serious research has been conducted.
- 5. As mentioned earlier, research may also be presented at academic conferences or published within academia. Participants will be identified in the projects by their real names.
- 6. Research may also be presented at academic conferences or published within academia. Participants will be identified in the projects by their real names due to the nature of the research. The reason for this is that the community is small and people know each other, thus maintaining confidentiality may not be possible.

- 7. There is no compensation for participation in this study.
- 8. If you have any questions regarding the research, please contact the researcher, Ambrita Nand at (831) 431-9339, or by e-mail, ambritak@gmail.com. Complaints about the research may be presented to the Anthropology Department Chair Chuck Darrah. He can be reached at (408)924-5314, or by e-mail Chuck.Darrah@sjsu.edu. Questions about a research subjects' rights, or research-related injury may be presented to Pamela Stacks, Ph.D., Associate Vice President, Graduate Studies and Research, at (408) 924-2427.
- 9. No service of any kind, to which you are otherwise entitled, will be lost or jeopardized if you chose not to participate in the study.
- 10. Your consent is being given voluntarily. You may refuse to participate in the entire study or in any part of the study, which consists of interviews and follow-up interviews. You have the right to not answer questions you do not wish to answer. If you decide to participate in the study, you are free to withdraw at any time without any negative effect on your relations with San Jose State University.
- 11. At the time that you sign this consent form, you will receive a copy of it for your records, signed and dated by the investigator.
- * The signature of a subject on this document indicates agreement to participate in the study.
- * The signature of a researcher on this document indicates agreement to include the above named subject in the research and attestation that the subject has been fully informed of his or her rights.

I have read the following information provided above and agree to participate in this research.

| Agree to audio recordings for the in | terview |
|--------------------------------------|---------|
| PARTICIPANT | |
| Print name: | |
| Signature: | |
| Date: | |
| INVESTIGATOR | |
| Print name: | |
| Signature: | |
| | |

Date:

Appendix B: Interview Protocol and Sample Questions

Interview Protocol

Welcome and thank you for your participation today in this interview. My name is Ambrita Nand and I am a graduate student at San Jose State University. I am conducting my study in partial fulfillment for the Masters of Applied Anthropology. This interview will take about 60 minutes and will include a series of questions regarding your experiences with being a first-generation Indo-Fijian woman residing in California's Central Valley, specifically in Modesto. I would like your permission to tape-record this interview so that I may accurately document our conversation. If at anytime you would like to discontinue the use of the tape-recorder or the interview itself, please feel free to let me know, as it is important that you feel comfortable. Your responses are important to this study and they will be used to gain a better understanding of how immigration affects the first-generation as well as the second-generation.

At this time I would like to take the time to remind you of your (written or oral) consent to participate in this study. I am the responsible investigator, specifying your participation in this research.

Read only if written consent is given:

You and I have both signed and dated each copy, verifying that we agree to do this interview. You will receive one copy and I will keep the other copy, which will be securely stored and locked away.

Also, your participation in this interview today is completely voluntary. If at any time you need to stop or take a break, once again, please feel free let me know. You may also withdraw your participation at any time without any consequence. Do you have any questions or concerns before we begin the interview?

(Answer all necessary questions or concerns)

Now, with your permission, we will begin the interview.

LIST OF SAMPLE INTERVIEW QUESTIONS

Interview Questions for First-Generation Participants

Interview will take approximately 60 minutes. These are open-ended questions. Additional questions may be asked to help answer the research questions.

- 1. What is your name and age? In what town or city were you born? What country? How old were you when you first arrived to the United States? What year was that? When did you first arrive in Modesto? What year was that?
- 2. Describe your homeland. What language did/do you speak there? How did your community differ from where you live today? What foods did you eat? Did people dress different there than they do in the United States? Where people religious where you came from? Are you religious? What types of traditions did you have in your homeland? What kind of job did you

have there? What do you miss about your homeland? Generally, what was your life like there? What was going on historically? Can you tell me about the government in your native country?

- 3. Why did you and your family and/or your family leave your native country? Describe your trip to the United States. What obstacles did you have to overcome? What experiences did you have? Any memorable experience(s)?
- 4. Before you came, what did you think life in the United States would be like? What had you heard about the United States? How you think you life would be different in the United States as compared to it in your homeland? Had you seen pictures of the U.S. before you came? What did you think of them? What had people told you about the United States? What did you hear about women in the United States?
- 5. Once you arrived in the United States, what did you experience? Was it different from what you expected? Explain. What stereotypes/expectations did you have? Did you experience any "culture shock"? What possessions did you bring with you? What important things, if anything, did you leave behind? What specific cultural traditions did you bring with you?
- 6. What kind of challenges did you experience adjusting to life and culture in the United States? Know or did you have to learn English? Did you change your eating habits? Did you change the way you dress? Did you change the way you related to your friends and family? Did or do you miss family members/friends? Did you change the way you learn? Did you change your leisure activities? Has life improved since immigration?
- 7. What differences have you noticed in regard to: living circumstances (housing, food, transportation, etc.), education, jobs, gender roles (what is expected of men or women), social customs and culture, values and attitudes? What are some customs that you have felt comfortable keeping? Have you dropped any customs because you felt some pressure to do so?
- 8. Do you have a green card? How did you obtain it? Are you an official U.S. citizen? When did you become an official citizen? How did you end up in Modesto, CA? Did you live anywhere else before Modesto, CA? Did you know anyone when you arrived in the United States? In Modesto?
- 9. What kind of job(s) did you have when you came to the United States? How did you find it? What was your experience? What do you do know for a living? Do you think it is easy for you as an Indo-Fijian woman to find work?
- 10. Did you go to school in the United States? What was your school experience like? Did you find school or English challenging? What challenges has learning English created for you? Do your parents speak English? If not, has this created any communication problems? What language do your children speak? Do you think you will ever forget your native language? How does that make you feel?

- 11. Have you faced discrimination in the United States based on your women, race, religion, or culture? If, have ever had a similar problem in your homeland? How did the experience(s) in this country make you feel? What did you do to overcome prejudice? What advice would you give to new immigrants from your country about this problem?
- 12. What do you believe it mean to "be an American"? Do you feel mostly American now or something else? What you believe in means to be Indo-Fijian? What do you know about Indo-Fijian history? What do you think life would be like for you if you did not migrate?
- 13. Are you married? When and where did you get married? Is your spouse Indo-Fijian? If not, what does your spouse know about your native homeland and culture? If you are single do you have plans for finding an Indo-Fijian mate?
- 14. If you have children where they born in this country? Do they speak your native language? Do they know about your immigration experience/native culture? How might your children's life differ if you did not emigrate? What traditions and values have you passed to them?
- 15. Do you ever visit Fiji? Do you have plans of returning someday?
- 16. Is there anything that I haven't asked you or that you would like to talk about that was not addressed in the interview?
- 17. Would it be okay to contact you if I have any additional follow-up questions?

Interview will take approximately 60 minutes. These are open-ended questions. Additional questions may be asked to help answer the research questions.

- 1. What is your name and age? When were you born and in what town? Describe your childhood? Describe your parents and your brothers and sisters if you have any? Do you know any stories about how your family first came to the United States? Where did they first settle? Why? How did they make a living? Did your family stay in one place or move around? How did they come to live in this area? What stories have come down to you about your parents and grandparents? More distant ancestors?
- 2. What year did you move to Modesto, CA? How long have you been living here? Where are other places that you have lived?
- 3. What languages do you speak? Which language did you speak at home with your parents? Where did you go to school? Describe your school experience(s) in elementary, junior high, high school, and college if any. Did you parents help you with school homework? Did you enjoy school? Was it difficult or easy? What did you want to be when you were a child?
- 4. What kinds of food did you eat growing up? What kinds of foods do you eat now? What about leisure and activities? Did you watch Bollywood films? Do you still?

- 5.Are you religious? What types of cultural traditions did you grow up with? What types of cultural traditions do you practice now? Did you go to any events related to your parent's native culture? Do you hang out with other Indo-Fijians? If yes, when and where? Do you wear traditional clothes? Do you use traditional items and cook traditional foods for them?
- 6. What was your first job? What other jobs have you held? What jobs did you like the best and which one's did you like the least? What were your experiences like?
- 7. Have you faced discrimination in the United States based on your women, race, religion, or culture? How did the experience(s) make you feel? What did you do to overcome prejudice? What advice would you give to others about this problem?
- 8. What do you believe it mean to "be an American"? Do you feel "Americanized"? Do you feel mostly American now or something else? What you believe in means to be Indo-Fijian and Indo-Fijian American?
- 9. Are you married? When and where did you get married? Is your spouse Indo-Fijian? If not, what does your spouse know about your native homeland and culture? If you are single do you have plans for finding an Indo-Fijian mate?
- 10.If you have children where were they born? Do they speak your native language? Do they know about your immigration experience/native culture? How might your children's life differ if you did not emigrate? What traditions and values have you passed to them? Do you stay home with your kids or watch your own kids when you are at work?
- 11.Do you ever visit Fiji? Do you have family there? Where else do you have family? Do you have plans of visiting or living there someday?
- 12. Is there anything that I haven't asked you or that you would like to talk about that was not addressed in the interview?
- 13. Would it be okay to contact you if I have any additional follow-up questions?